The copyright © of this thesis belongs to its rightful author and/or other copyright owner. Copies can be accessed and downloaded for non-commercial or learning purposes without any charge and permission. The thesis cannot be reproduced or quoted as a whole without the permission from its rightful owner. No alteration or changes in format is allowed without permission from its rightful owner.



# A STUDY ON THE RELATIONSHIP BETWEEN WORK FACTORS AND EMPLOYEE SATISFACTION



Thesis Submitted to Othman Yeop Abdullah Graduate School of Business Universiti Utara Malaysia in Partial Fulfillment of the Requirement for the Master of Human Resource Management



Othman Yeop Abdullah Graduate School of Business

Universiti Utara Malaysia

#### PERAKUAN KERJA KERTAS PENYELIDIKAN (Certification of Research Paper)

Saya, mengaku bertandatangan, memperakukan bahawa (I, the undersigned, certified that) SUHANA BT MOHD NAJID (818755)

Calon untuk Ijazah Sarjana (Candidate for the degree of) MASTER OF HUMAN RESOURCE MANAGEMENT

Telah mengemukakan kertas projek yang bertajuk (has presented his/her project paper of the following title)

#### A STUDY ON THE RELATIONSHIP BETWEEN WORK FACTORS AND EMPLOYEE SATISFACTION

Seperti yang tercatat di muka surat tajuk dan kulit kertas projek (as it appears on the title page and front cover of the project paper)

Universiti Utara Malavsia

Bahawa kertas projek tersebut boleh diterima dari segi bentuk serta kandungan dan meliputi bidang ilmu dengan memuaskan.

(that the project paper acceptable in the form and content and that a satisfactory knowledge of the field is covered by the project paper).

NamaPenyelia (Name of Supervisor) •

ċ

NORIZAN BINTI, HAJI AZIZAN

Tandatangan (Signature)

Tarikh *(Date)*  2 AUGUST 2017

#### PERMISSION TO USE

In presenting this dissertation/project paper in partial fulfilment of the requirements for a Post Graduate degree from Universiti Utara Malaysia (UUM), I agree that the Library of this university may make it freely available for inspection. I further agree that permission for copying of this dissertation/project paper in any manner, in whole or in part, for scholarly purposes may be granted by my supervisor(s) or, in their absence, by the Dean of Othman Yeop Abdullah Graduate School of Business where I did my dissertation/project paper. It is understood that any copying or publication or use of this dissertation/project paper part of it for financial gain shall not be allowed without my written permission. It is also understood that due recognition shall be given to me and to UUM in any scholarly use which may be made of any material in my dissertation/project paper.

Request for permission to copy to make other use of materials in this dissertation/project paper in whole or in part should be addressed to:



#### ABSTRACT

Employee satisfaction enhances confidence, loyalty and consequently improved work quality, productivity and organizational performance. Various factors such as benefits and services provided for employees, health insurance, welfare, rewards as well as training can affect employee satisfaction. The quality level of services provided to employees can influence employee satisfaction. One of the benefits provided by the company in this study is housing loan. Nonetheless, many employees are not satisfied with the quality service provided. Many complaints were lodged with the company on the dissatisfaction. As such, this study was undertaken to investigate the relationship between employee satisfaction and quality service from the aspects of tangible, reliability and responsiveness. A questionnaire on service quality was utilised to examine the relationship between variables. 92 respondents participated in this study. Results showed that all there independent variables measuring service quality were positively and significantly related to employee satisfaction. Reliability of services provided was found to be the most critical towards employee satisfaction. As a conclusion, the company is strongly recommended to review relevant policies and procedures relating to quality of services provided to employees in order to increase employee satisfaction.

Keyword: Employee Satisfaction, Tangible, Reliability, Responsiveness.

Universiti Utara Malavsia

iii

#### ABSTRAK

Kepuasan pekerja dapat mewujudkan keyakinan, kesetiaan and secara tidak langsung akan meningkatkan hasil kualiti kerja, produktiviti dan prestasi organisasi. Pelbagai faktor seperti faedah dan perkhidmatan yang diberi, insuran kesihatan, aspek kebajikan serta ganjaran dan latihan dapat mempengaruhi kepuasan pekerja. Kualiti terhadap perkhidmatan yang disediakan untuk pekerja juga dapat mempengaruhi kepuasan pekerja. Salah satu faedah yang disediakan oleh organisasi yang dikaji untuk pekerja adalah pinjaman peumahan. Walaubagaimanapun, ramai pekerja yang tidak berpuashati dengan kualiti perkhidmatan tersebut. Banyak aduan yang diterima oleh pihak organisasi. Oleh yang demikian, kajian ini dilaksanakan bagi melihat perhubungan antara kepuasan pekerja dengan kualiti perkhidmatan yang diberikan dari aspek kebolehnyataan, kebolehpercayaan dan tanggapan. Soalselidik berkaitan kualiti perkhidmatan digunakan bagi menguji hubungkait ini. Seramai 92 responden terlibat dalam kajian ini. Dapatan kajian menunjukkan bahawa ketiga-tiga pembolehubah mempunyai hubungan yang signifikan dan positif dengan kepuasan pekerja. Dapatan menunjukkan bahawa kebolehpercayaan terhadap perkhidmatan yang disediakan adalah faktor yang penting untuk kepuasan pekerja. Sebagai rumusan, organisasi dicadangkan agar melihat semula dasar dan prosedur tentang kualiti perkhidmatan yang disediakan untuk pekerja agar dapat meningkatkan kepuasan pekerja secara keseluruhannya.

Kata kunci: Kepuasan pekerja, Kualiti Perkhidmatan, Kebolehnyataan, Kebolehpercayaan

Universiti Utara Malaysia

#### ACKNOWLEDGEMENT

First and foremost, all praise to Allah the most gracious and most merciful for His Blessing for giving me the inspiration to embark on this study and instilling in all of my strength to see that this research become a reality. Blessing and salutation also be on Prophet of Allah SWT, Muhammad SAW.

I would like to take this opportunity to express my gratitude and appreciation to my supervisor Pn. Norizan bt Hj Azizan. I am truly indebted for her contribution in term of time, patience, efforts, encouragements, knowledge, tolerance and advices throughout the process of completing this project paper.

A great thanks to the staff of XYZ Berhad and headquarters because giving me a great opportunity and cooperation to conduct my study in their organisation.

#### Jniversiti Utara Malaysia

A special thanks to my beloved parents, Mohd Najid bin Mohamad and Fauziah bt Mat Zain for their demonstration of love and continuous moral supports throughout my years of study.

Finally, i would like to convey my thanks to my special friends which has been an inspiration for me to move forward; Dzafarul Idham bin Rubaidin, Noorimah bt Mohd Mustaffa and Liyana bt Aziz.

## **Table of Contents**

Declaration	
Permission to Use	ii
Abstract	iii
Acknowledgement	v
Table of Content	vi
List of Appendix	xi
List of Table	xii
List of Figure	xiii

## **CHAPTER 1: INTRODUCTION**

1.1	Introduction	1
1.2	Background of the study	1
1.3	Background of the organization	2
1.4	Problem Statement	
1.5	Research Questions	4
1.6	Research Objectives	5
1.7	Significance of Study	5
1.8	Scope and Limitation of Study	6
1.9	Definition of Key Terms	6
1.10	The organization of thesis	8

## **CHAPTER 2: LITERATURE REVIEW**

2.1	Introduction	0
2.2	Employee Satisfaction1	0

2.3	Tangible
2.4	Reliability2
2.5	Responsiveness
2.6	Relationship between Independent Variables and Dependent Variables23

## **CHAPTER 3: METHODOLOGY**

3.1	Introduction	28
3.2	Research Framework	28
3.3	Research Design	29
3.4	Operational Definition	30
3.5	Measurement of Variables	31
3.6	Questionnaire Design	33
3.7	Data Collection	35
	3.7.1 Population and Sampling Design	36
	3.7.2 Sampling Technique	37
	3.7.3 Technique of Data Analysis	38

## **CHAPTER 4: RESULTS AND DISCUSSION**

4.1	Introduction	42
4.2	Response Rate	42
4.3	Profile of Respondents	42
4.4	Descriptive Analysis	49
4.5	Reliability Analysis	54

.

4.6	Pearson Correlation Testing	55
4.7	Discussion	57
4.8	Multiple Regression Coefficient	58
4.9	Summary	60

### **CHAPTER 5: CONCLUSION AND RECOMMENDATION**

5.1	Introduction	61
5.2	Summary of Study	61
5.3	Discussion	62
5.4	Implication of the Study	64
5.5	Recommendation	65
5.6	Direction for Future Research	69
5.7	Conclusion Universiti Utara Malaysia	70

## REFERENCES

APPENDICES

## LIST OF TABLE

## Table

## Page Number

Table 3.1	Summary Sources of Researched Variable Measurements	
Table 3.2	Summary of the Questionnaire	
Table 3.3	Statistic of Respondents	36
Table 3.4	Population of Respondents	36
Table 3.5	Table Krejcie and Morgan	37
Table 3.6	Interpreting Cronbach's Alpha Value	39
Table 3.7	Scale Rating by Hair, Money, Samouel and Page	41
Table 4.1	Demographic of Respondents	43
Table 4.2	Category of Mean Score	50
Table 4.3	Descriptive Statistics of Tangibles	50
Table 4.4	Descriptive Statistics of Reliability	56
Table 4.5	Descriptive Statistics of Responsiveness	52
Table 4.6	Descriptive Statistics of Employee Satisfaction	53
Table 4.7	Reliability of Variables	54
Table 4.8	Interpreting the Cronbach's Alpha Value	55
Table 4.9	Correlation Coefficient	55
Table 4.10	Scale rating developed by Hair, Money, Samuel and Page	56
Table 4.11	Multiple Regression Coefficient	58

### LIST OF FIGURE

## Figure

## Page Number

Figure 3.1	Theoritical Framework	28
Figure 4.1	Statistics Gender of Respondents	44
Figure 4.2	Statistics Age of Respondents	45
Figure 4.3	Statistics of Work Sector	45
Figure 4.4	Statistics of Job Category	46
Figure 4.5	Statistics Length of Service	47
Figure 4.6	Frequency of Communication	48
Figure 4.7	Frequency Method of Communication Universiti Utara Malaysia	49

#### **CHAPTER 1**

#### INTRODUCTION

#### 1.1 INTRODUCTION

The first chapter will explain on background and the focus of the study. The discussion in this chapter is organized in the following manner: (i) background of the study, (ii) problem statement, (iii) research questions, (iv) research objectives, (v) significance of study, (vi) scope and limitations of the study, and (vii) definition of key terms.

The purpose of this research is to examine the relationship between employee satisfaction and the following factors: tangible, reliability and responsiveness.

Universiti Utara Malaysia

Employee satisfaction is essential for business success in today's marketplace. It measures expectations of employee in regard of a product and service provided by the company. Employee satisfaction is defined by the number of employees or percentage of total employees, who reported their experiences with a firm, its products or its services (rating) against specified satisfaction measures (Nippatlapalli, 2013).

According to Khafafa & Shafii (2013), employee satisfaction is one of the most important key factors for any organisational survival. Employee satisfaction is an effective part of a business fulfilment. In addition, business would not be able to

survive with dissatisfied employees (Rahman *et al.*, 2012). Most successful organisations have the following elements: high level of employee satisfaction, ability to attract new clients and maintain old clients. These are the main essence of a successful organisation (Yuksel, 2010).

The relationship between employee and an organization is influenced by the quality of service given by an organization. In view of that, maintaining existing employee through the improvement of employee satisfaction and quality service should be done constantly as a way to achieve organization's targeted profit.

#### **1.2 BACKGROUND OF THE STUDY**

XYZ Berhad is one of Subsidiaries Company of XYZ Headquarters which was incorporated on 15 November 1994, under Section 16 (4) of the Companies Act 1985 with a paid-up capital of RM8.4 million and authorized capital of RM25 million. Its major business activities including managing funds of headquarters, factoring and financing; personal loan, motor vehicle loan and housing loan. It is also responsible in administering the activities of *Rancangan Tanah Baru Berkelompok* (RTBB) cash components investment funds of headquarters.

The most important department in the company is Housing Loan Department. It is managed by dedicated team known as *Akaun Pembiayaan* (AP). The main function is to provide housing facility for the eligible staff of PRB including staff based in headquarters.

During the early days of establishment of XYZ Berhad, there is no system provided. Most of the activities are managed manually by the dedicated team until the BIARIS system (*Sistem Pembiayaan*) being introduced in year 2000. It is undeniable that during the period, XYZ Berhad faced difficulties in term of managing and processing the housing loan applications.

#### **1.3 PROBLEM STATEMENT**

XYZ Berhad (which will be referred to as "the company" in this study) has been operating for about 17 years to provide housing loan for staff. It performs the same essential functions as other banks. However, it is difficult to standardize the services due to its tangibility and also due to lack of qualified and experienced manpower which ultimately translate into low quality service.

Since this housing loan facility is only provided by the company and not by other subsidiaries, eligible employees seize the opportunity without taking into consideration the quality of the services provided. Thus, this situation finally invites employees' dissatisfaction.

The company receives many complaints in regards of services. Some of the complaints included lost data due to improper filing system, inefficient staff handling those loans, as well as poor customer service given. Sometimes loan enquiries took too long for the company to respond to. Responsiveness rate was too long that most employees were unhappy with. Reliability was another issue related to lost data as enquiries could not be addressed immediately. This could also be

attributed to poor training, poor filing system or staff attitude. There are various mediums for employees to voice their dissatisfaction such as fax, phone calls and written complaints. All complaints are recorded for further action.

There are cases whereby some of the employees prefer to make complaints directly to the CEO without following the standard procedure. Normally these cases are considered as unrecorded complaints and difficult for the person in charge to follow up the cases due to improper notification.

Due to the growing number of eligible employee and also number of complaints, it is imperative for the company to take an initiative to improve the level of employee satisfaction by increasing the quality of services provided. On that note, this study was conducted to investigate the factors which affected employee satisfaction, namely tangibles, reliability and responsiveness.

🖉 Universiti Utara Malavsia

#### 1.4 RESEARCH QUESTIONS

Based on the problem highlighted and discussed above, the research questions will be addressed in this study:

- 1. Is there any significant relationship between tangible and employee satisfaction?
- 2. Is there any significant relationship between reliability and employee satisfaction?

3. Is there any significant relationship between responsiveness and employee satisfaction?

#### **1.5 RESEARCH OBJECTIVES**

- 1. To determine the relationship between tangible and employee satisfaction
- 2. To determine the relationship between reliability and employee satisfaction
- 3. To determine the relationship between responsiveness and employee satisfaction

#### **1.6 SIGNIFICANCE OF STUDY**

Generally, this study aims to understand the current work factors and employees' satisfaction in the company. Given direct participation from employees in this study, it provides first hand feedback and meaningful information for the company regarding employees' perceptions on their services.

The results will be able to guide the company to set their strategies in improving employees' satisfaction towards work factors. The management can use the inputs to help improve relevant policies and procedures, as well as delivery system, in order to improve their service to the customers who are the employees themselves.

The findings of this study can provide useful information for managers and researchers in the field related to employees' satisfaction and work factors.

#### 1.7 SCOPE AND LIMITATION OF STUDY

The study is based on primary data collected through questionnaire from respondents. Due to time limitation, this study involved only one company. Another challenge faced by the researcher was when some of the expected respondents were reluctant to complete the questionnaires distributed.

Secondly, this study only involved employees from the headquarters in Kuala Lumpur. The headquarters was chosen as it has the largest number of employees served by the company.

Thirdly, the researcher faced difficulties in getting the actual recorded data of the complaints and details of the complaints received from the management due to private and confidential issues.

#### **1.8 DEFINITION OF KEY TERMS**

In this section, working definitions of the variables in this study are described briefly. The two of main subjects which is employee satisfaction and work factors will be explained in this part. These definitions are to ensure a common understanding of the terms throughout this research. More detailed definitions will be provided in the following chapter.

#### 1.8.1 Employee

A person who works part-time or full time under a contract of employment for wages or salary and in a position below the executive level.

#### 1.8.2 Satisfaction

Satisfaction refers to the situation where people feel their perception exceeds the expectation.

#### 1.8.3 Employee Satisfaction

Employee satisfaction demonstrates the perceived relationship between the expectation of an employee from his job and his perceived offerings of the job (Locke, 1976).

#### 1.6.4 Tangible

Tangibles include appearance of physical facilities, equipment, personnel and communication materials involved in the delivery of service to employees.

#### 1.6.5 Reliability

Reliability can be referred to as the ability to perform promised services which are dependable with accuracy.

#### 1.6.6 Responsiveness

The willingness to help and provide prompt service, responsiveness concerns the willingness or readiness of employee's to provide service. It involves timeliness of service.

#### 1.9 ORGANISATION OF THESIS

This report is organised into five main chapters.

Chapter 1 provides the background of the study. It highlights the focus of the study. This chapter includes discussions on problem statement, research questions, research objectives, significance of the study, followed by scope and limitation of study.

## Universiti Utara Malaysia

Chapter 2 discusses past literature related to employee satisfaction and factors that might contribute to employee satisfaction specifically on tangible, reliability and responsiveness.

Chapter 3 discusses the research methodology and techniques utilised in this study. This includes the research framework, research design, population of the study, sampling method and data collection procedures. Chapter 4 presents findings from data gathered. Reliability of the instrument is presented using Cronbach's coefficient alpha. Data analysis was generated using Statistical Package Social Science (SPSS) version 21.0.

Chapter 5 presents the discussion of the findings and its summary, and concludes with suggestion for future research.



#### CHAPTER 2

#### LITERATURE REVIEW

#### 2.1 Introduction

This chapter reviews past studies relating to the study. Literature relating to the variables involved, namely employee satisfaction as the dependent variable and tangible, reliability and responsiveness as the independent variables, are reviewed.

#### 2.2 Employee Satisfaction

Employee satisfaction refers to how content or satisfied employees are with their jobs (Griffin & Moorhead, 2013). It is typically measured using an employee satisfaction survey. An employee satisfaction survey basically addressed employees' perceptions on compensation, workload, management, flexibility, teamwork and resources. Islam and Siengthai (2009) perceived employee satisfaction as a positive emotional state resulting from the appraisal of one's job or job experiences.

Employee satisfaction also can be defined as the terminology used to describe whether employees are happy, contended and fulfilling their desires and needs at work. Many measures support that employee satisfaction is a factor in employee motivation, employee goal achievement and positive employee morale in the work place (Susan M.Heath field, 2011). Meanwhile, according to Nancy C. Morse (1997), satisfaction refers to the level of fulfilment of one's needs, wants and desire. It depends basically upon what an individual wants from the world, and what he gets. Lai Wan (2007), satisfaction is an important goal for organisations to reach, as it has been shown that profitability, productivity, employee retention and customer satisfaction are linked to employees' satisfaction. Employee satisfaction is a measure of how happy workers are with their job and working environment. He believed that there are many factors affecting the organizational effectiveness and one of them is the employee satisfaction.

"Employee satisfaction also known as job satisfaction, is the extent to which an individual is happy with their job and the role it plays in their life", (HR Zone). This is in line with Locke (1976), which stated that employee satisfaction known as job satisfaction. It is a positive emotional state that demonstrates the perceived relationship between the expectation of an employee from his job and his perceived offerings of the job. Spector (1997), defined employee satisfaction is a comprehensive term that comprises job satisfaction of employees and their satisfaction overall with companies' policies, company environment

Price (2001) defines employee satisfaction as the effective orientation that an employee has towards his or her work. Sempane, Rieger and Roodt, (2002) recognised employees' satisfaction as the individual's perception and evaluation of the overall work environment. As for Lu, While and Barriball (2005), employee

satisfaction is a global feeling about one's work or a related cluster of attitudes about various facets of the work environment.

Griffin and Moorhead (2013), described employee satisfaction as the feeling of gratification or prosperity that employees procure from their job; whether they are happy to work or not, perceive their jobs as meaningful, or the extent to which their job has a negative physical/psychological effect on them. Employee satisfaction as simply how people feel about their jobs and various aspects of their jobs (Lawler 1990).

Employee satisfaction is the fulfilment or satisfying emotional state that results from the positive appraisal of job experiences on the part of the employee (Chang, C., Chiu, C. and Chen, C. 2010; Karatepe, 2012; Al-Refaie, 2015). Meanwhile, as according to Chi and Gursoy, 2009; Karatepe, O.M., Uludag, O., Menevis,I., Hadzimeh medagic, L and Baddar, L (2006), they found that employee satisfaction is a person's perception or appraisal of the degree of fit between an individual and the organization's values. Hsu and Wang, 2008; Prajogo and Cooper, 2010; Jung and Yoon, 2015 defines employee satisfaction as a person's assessment of the overall quality of his or her current job assignment.

Employee satisfaction is defined as the combination of affective reactions to the differential perceptions of what he/she wants to receive compared with what he/she actually receives. Therefore, the organizations should try to supply the employee

expectations in order to approach the employee satisfaction. In addition, emotional state of the employees may also affect their satisfaction. This forces the managers to create and sustain the desired working environments in the organizations. (Cranny, Smith, & Stone, 1992)

#### 2.2.2 Previous Studies on Employee Satisfaction

The employee satisfaction described as how pleased an employee is with his or her position of employment (Moyes, Shao, & Newsome, 2008). It is recognized as one of the most important drivers of employee service quality, productivity and loyalty (Matzler and Renzl, 2006). Employee satisfaction can be related to firm performance in terms of growth. The practices of managing human resources show that the possibility of education, adequate pay, benefits, continuity of employment and the right approach to employees encourage a high level of employee motivation and their willingness to invest in their own knowledge and skills (Shaw, J.D., Delery, J.E., Jenkins, G.D. Jr and Gupta, N. 1998; Tsui, A.S., Pearce, J.L., Porter, L.W. and Tripoli, A.M., 1997). (1997). For most management scientists, meeting the needs of employees remains the prime employee satisfaction-enhancement strategy (Giannikis & Mihail, 2011)

Hunter & Tietyen, 1997 conclude that employees are more loyal and productive when they are satisfied and thus affect the organizational productivity (Potterfield, 1999). Ellickson and Logsdon (2001), gives emphasis to environmental factors and

personal characteristics as the two most influential variables that determine the level of employee satisfaction.

Improving the satisfaction of workers is a central task of management since satisfaction creates confidence, loyalty and consequently improved quality in the output of employees (Tietjen and Myers, 1998). In order to improve employee satisfaction, it is important to measure and establish the existing levels first (Wright, Gardner, Moynihan & Allen, 2005).). Top management's commitment to improving employee satisfaction takes into account factors that affect employee satisfaction and can encourage employees to improve the performance of their tasks and boost the level of their work performance, which can in turn contribute to the company's growth (Tsui *et al.*, 1997; Shaw *et al.*, 1998; Gerhart and Rynes, 2003).

## Universiti Utara Malaysia

Organ and Ryan (1995), stated that the employee satisfaction is one of the basics of organizational citizenship behaviour. The well-satisfied employees will work more willingly and this contributes to the effectiveness of their organizations. Highly satisfied employees demonstrate higher level of loyalty to the organization (Wagner and Herter, 2006).

Previous studies have proposed that employees are the greatest assets of a company, and that satisfied customers must satisfy employee requirements (Nebeker, D., Busso, L., Werenfels, P.D., Diallo, H., Czekajewski, A. and Ferdman, B. 2001), According to Lee (1988), employee satisfaction is as important

as customer satisfaction in influencing organisational performance. It showed that job satisfaction is among the best predictors of turnover.

Employee satisfaction can improve productivity, reduce staff turnover and enhance creativity and commitment. Therefore, employee satisfaction should not be ignored and yet very few businesses seriously consider employee satisfaction (Ulmer, D., Syptak, J.M. and Marsland, D.W. 1999).

The role of employee training and the top management leadership of employees can be essential for the quality and performance of firms (Demirbag, M., Koh, S.C.L., Tatoglu, E. and Zaim, S., 2006). Meanwhile, Demirbag *et al.*, 2006).

## Universiti Utara Malaysia

Moyes, Shao and Newsome (2008), found that in order to increase employees' satisfaction, the organisation must have good relationships with the colleagues, high salary, good working conditions, training and education opportunities, career developments or any other benefits may be related with the increasing of employees' satisfaction.

When investigating the employees' satisfaction, it should be known that an employee may be more satisfied by a satisfying item, whereas the other employee may be less satisfied with the same item. Because of this, analyzing the employee satisfaction from a large perspective will be better. That means; the sum of all satisfying factors composes that employee's satisfaction level. As a general definition, the employee satisfaction may be described as how pleased an employee is with his or her position of employment (Moyes, Shao & Newsome, 2008)

There are many factors that influence employee satisfaction across industries and countries. The most important are wage structure, working conditions, work group, the nature of the work and the quality of supervision and salary (Sousa-Poza & Sousa-Poza, 2000; Jun, Cai and Shin (2006); Rogelberg, Allen, Shanock, Scott & Shuffler, 2010). Training and career development plays an important role in employee satisfaction (Burke, Graham & Smith, 2005; Amin, Aldakhil, Wu, Rezaei, & Cobanoglu, 2017). Employee satisfaction is influenced by work–life balance (Ooi, *et al.* 2013; Lee, Back & Chan, 2015)

Rodriguez-Anton and Alonso-Almeida (2011), salary conditions (salary, salary complements, and job stability, among others), employee involvement, motivation in assigned tasks, improved occupational health and safety conditions and employee learning contribute to employee satisfaction.

The important factors influencing employee satisfaction are relationships with coworkers and supervisors. Thus, if employees perceive that these requirements are fulfilled, then this will enhance the level of employee satisfaction (Rogelberg *et al.* 2010). From a practical perspective, satisfied and motivated employees will remain with organizations because they perceive more benefits in continuing to working with these organizations (Koyuncu *et al.*, 2013). According to Eskildsen and Dahlgaard, (2000), satisfied workers work harder, are more motivated and have higher morale. Prior research also reveals that internal service quality helps employees to do their jobs better, which makes them feel more satisfied (Chiang & Wu, 2014; Nazeer, S., Zahid & Azeem, 2014; Pantouvakis, 2011).

Miller (2006) stated that there is no limit for the employees to reach the full satisfaction and it may vary from employee to employee. Sometimes they need to change their behaviours in order to execute their duties more effectively to gain greater job satisfaction.

## Universiti Utara Malaysia

Kelly (2005) emphasised the importance of understanding what the employees are satisfied by and measuring the employee satisfaction in the workplace as it is critical to the success, profitability and competitive advantage of the organization.

Bhatti and Qureshi, (2007), effective organizations should have a culture that encourages the employee satisfaction. Employees are more loyal and productive when they are satisfied (Hunter & Tietyen, (1997). This is because satisfied employees affect the customer satisfaction and organizational productivity Potterfield, (1999). Ganguly (2010) stated that maintaining that the person-environment fit paradigm has been widely recognised as the most appropriate explanation for employee satisfaction. Employee satisfaction is influenced by the interaction of a family of factors such as recognition, communication, co-workers, fringe benefits, working conditions, the nature of the work itself, the nature of the organisation itself, organisational systems, policies and procedures, compensation, personal development, promotion, appreciation, security, and supervision.

#### 2.2.3 Measurements of Employee Satisfaction

#### 2.2.3.1 Employee satisfaction in the field of higher education

Oshagbemi, (1997b), measured job satisfaction for 566 college teachers and identify the determinants of satisfaction in the field of higher education as shown below:

- i) teaching;
- ii) research;
- iii) administration and management;
- iv) present pay;
- v) promotions;
- vi) supervision/supervisor behaviour;
- vii) behaviour of co-workers; and

viii) Physical conditions/working facilities.

#### 2.2.3.2 Employee satisfaction in the field of private higher education

Comm and Mathaisel (2000), conduct questionnaire surveys on 606 employees of a private higher education organisation to identify the determinants of satisfaction within educational organisations. The findings were as follows:

- i) Workload;
- ii) Work atmosphere;
- iii) Decision-making;
- iv) Ethics/fairness;

#### 2.2.3.3 Employee satisfaction in the field of police organisation

Fosam, E.B., Grimsley, M.F.J. and Wisher, S.J. (1998), analysed police organisations to find a suitable employee satisfaction model taking the South Yorkshire Police (SYP) as an example. The determinants of satisfaction are as follows:

- i) Customer focus
- ii) Supervision;
- iii) Goals and objectives;
- iv) Training and development;
- v) Pay; and
- vi) Benefits.

#### 2.2.3.4 Employee satisfaction in Turkish College

Kusku (2001), proposed applying employee satisfaction surveys to the employees of a Turkish college, and applied the following dimensions for measuring their satisfaction:

- i) General satisfaction
- ii) Management satisfaction;
- iii) Colleagues;
- iv) Other working group satisfaction;
- v) Job satisfaction;
- vi) Work environment; and
- vii) Salary satisfaction.

#### 2.2.3.5 Employee satisfaction in public government sector

Metle (2003), conducted employment satisfaction surveys on female employees in the Kuwaiti public government sector (KGS), and identified the following employment satisfaction factors:

- i) Overall job satisfaction
- ii) Pay and security;
- iii) Co-workers;

- iv) Supervision;
- v) Promotion; and
- vi) Content of work.

### 2.3

No	Scholars	
1	Price, (2001)	Defines employee satisfaction as the effective orientation that an employee has towards his or her work.
2	Sempane, Rieger & Roodt, (2002)	Employee satisfaction recognised as the individual's perception and evaluation of the overall work environment
3	Lu, While and Barriball (2005)	Define employee satisfaction as a global feeling about one's work or a related cluster of attitudes about various facets of the work environment.
4	Islam & Siengthai, (2009)	Employee satisfaction perceived as a 'positive emotional state resulting from the appraisal of one's job or job experiences'
5	Ellickson and Logsdon, (2001)	Gives emphasis to environmental factors and personal characteristics as the two most influential variables that determine the level of employee satisfaction
6	Lambert, Edwards and Cabic (2003)	Found low employee satisfaction levels amongst employees whose expectations fell short

7	Ganguly (2010)	Maintains that the person-environment fit paradigm has been widely recognised as the most appropriate explanation for employee satisfaction.
8	Ilies, Wilson & Wagner (2009): Irving & Montes, (2009): Koonmee, Singhapakdi, Virakul & Lee (2010).	Employee satisfaction is influenced by the interaction of a family of factors such as recognition, communication, co-workers, tringe benefits, working conditions, the nature of the work itself, the nature of the organisation itself, organisational systems, policies and procedures, compensation, personal development, promotion, appreciation, security, and supervision.
9	Giannikis & Mihail, (2011)	For most management scientists, meeting the needs of employees remains the prime employee satisfaction- enhancement strategy
10	Locke (1976)	Employee satisfaction known as job satisfaction, is positive emotional state that demonstrates the perceived relationship between the expectation of an employee from his job and his perceived offerings of the job
11	Wagner and Herter (2006)	Advocate that highly satisfied employees demonstrate higher level of loyalty to the organization:
12	Chang, C., Chiu, C. and Chen, C. (2010); Karatepe, 2012; Al- Refaie, 2015	Employee satisfaction is the fulfilment or satisfying emotional state that results from the positive appraisal of job experiences on the part of the employee (belum)
13	Hsu and Wang, 2008; Prajogo and Cooper, 2010; Jung and Yoon, 2015	Employee satisfaction as a person's assessment of the overall quality of his or her current job assignment
14	Sousa-Poza and Sousa- Poza 2000, Jun, M., Cai, S. and Shin, H. (2006); Rogelberg, S.G., Allen, J.A., Shanock, L., Scott, C. and Shuffler, M. (2010)	There are many factors that influence employee satisfaction across industries and countries. The most important are wage structure, working conditions; work group, the nature of the work and the quality of supervision and salary.
15	Burke, R.J., Graham, J. and Smith, F. (2005); Amin, M., Aldakhil, A.	Training and career development plays an important role in employee satisfaction

	M., Wu, C., Rezaei, S., & Cobanoghi, C. (2017)	
16	Ooi, K.B., Lee, V.H., Chong, A.Y.L. and Lin, B. (2013), Lee, I.S., Back, K.J. and Chan, F.S. (2015)	Employee satisfaction is influenced by work-life balance
17	Chi and Gursoy, (2009); Karatepe, O.M., Uludag, O.,Menevis,I.,Hadzimeh medagic, L. and Baddar, L. (2006);	Employee satisfaction is a person's perception or appraisal of the degree of fit between an individual and the organization's values (Chi and Gursoy, 2009; Karatepe <i>et al.</i> , 2006).
18	Lai Wan (2007)	Satisfaction is an important goal for organisations to reach, as it has been shown that profitability, productivity, eniployee retention and customer satisfaction are linked to employees satisfaction
19	Rodriguez Anton and Alonso-Almenda (2011) +	Salary conditions (salary salary complements and job stability, among others), employee involvement, motivation in assigned tasks, improved occupational health and safety conditions and employee learning contribute to employee satisfaction.
20	Rogelberg, S.G., Allen, J.A., Shanock, L., Scott, C. and Shuffler, M. (2010)	The important factors influencing employee satisfaction are relationships with co-workers and supervisors. Thus, if employees perceive that these requirements are fulfilled, then this will enhance the level of employee satisfaction.
21	Koyuncu, M., Burke, R., Fixenbaum, L. and Tokus, Y. (2013)	From a practical perspective, satisfied and motivated employees will remain with organizations because they perceive more benefits in continuing to working with these organizations
22	Lawler, (1990)	Employee satisfaction as simply how people feel about their jobs and various aspects of their jobs.
23	Hunter & Tietyen, (1997); Potterfield, (1999) (IJBM ES)	Employees are more loyal and productive when they are satisfied (Hunter & Tietyen, 1997), and these satisfied employees affect the customer satisfaction and organizational productivity (Potterfield, 1999)

24	Cranny, Smith, & Stone, (1992) (IJBM ES)	Employee satisfaction is defined as the combination of affective reactions to the differential perceptions of what he/she wants to receive compared with what he/she actually receives (Cranny, Smith, & Stone, 1992). Therefore, the organizations should try to supply the employee expectations in order to approach the employee satisfaction. In addition, emotional state of the employees may also affect their satisfaction. This forces the managers to create and sustain the desired working environments in the organizations.
25	Organ and Ryan (1995) (IJBM DS)	Organ and Ryan (1995) stated that the employee satisfaction is one of the basics of organizational citizenship behaviour. The well-satisfied employees will work more willingly and this contributes to the effectiveness of their organizations
26	Miller (2006).	There is no limit for the employees to reach the full satisfaction and it may vary from employee to employee. Sometimes they need to change their behaviours in order to execute their duties more effectively to gain greater; job satisfaction.
27	Moyes, Shao, & Newsome, (2008). (IJBM ES)	The employee satisfaction described as how pleased an employee is with his or her position of employment
28	Kelly, (2005) (IJBM ES)	Investigating what the employees are satisfied by and measuring the employee satisfaction in the workplace is critical to the success and increases the profitability of the organization for having competitive advantage
29	Nancy C. Morse, (1997)	Satisfaction refers to the level of fulfillment of one's needs, wauts and desire. It depends basically upon what an individual wants from the world, and what he gets. Employee satisfaction is a measure of how happy workers are with their job and working environment. It is sure that there may be many factors affecting the organizational effectiveness and one of them is the employee satisfaction.
30	Bhatti & Qureshi, (2007)	Effective organizations should have a culture that encourages the employee satisfaction
31	Hunter & Tietyen, (1997)	Employees are more loyal and productive when they are satisfied
32	Potterfield, (1999)	Satisfied employees affect the customer satisfaction and organizational productivity
---	--	--
33	Spector (1997)	Employee satisfaction is a comprehensive term that comprises job satisfaction of employees and their satisfaction overall with companies' policies, company environment
34	Griffin and Moorhead, (2013)	Employee satisfaction is generally described as the feeling of gratification or prosperity that employees procure from their job; whether they are happy to work or not, perceive their jobs as meaningful, or the extent to which their job has a negative physical/psychological effect on them.
		http://www.emeraldinsight.com.eserv.uum.edu.my/doi/f ull/10.1108/JOSM-10-2015-0294
35	Matzler and Renzl. (2006)	Employee satisfaction is recognized as one of the most important drivers of employee service quality, productivity and loyalfy http://www.emeraldinsight.com.esery.uum.edu.my/doi/f ull/10.1108/JOSM-10-2015-0294
36	Eskildsen and Dahlgdard.	Satisfied workers work harder, are more motivated and
A contra de la contr	<u>(2009)</u>	have higher morale http://www.emeraldinsight.com.eserv.uum.edu.my/doi/f- ull/10.1108/JOSM-10-2015-0294
37	Chiang and Wu, (2014); Nazeer, S., Zahid, M.M. and Azeem, M.F., (2014); Pantouvakis, (2011)	Prior research also reveals that internal service quality helps employees to do their jobs better, which makes them feel more satisfied http://www.emeraldinsight.com.eserv.uum.edu.my/doi/f ull/10.1108/JOSM-10-2015-0294
38	Wright, Gardner, Moyuihan & Allen, (2005). (ISQ-ES)	In order to improve employee satisfaction, it is important to measure and establish the existing levels first
39	Dubrovski, (2001)	Improving eustomer satisfaction not only raises company profits, but also facilitates company development. Employee satisfaction influences organisational performance as much as customer

		satisfaction.
40	Nebeker, D., Busso, L., Werenfels, P.D., Dialfo, H., Czekajewski, A. and Ferdman, B. (2001), "Airline station performance as a function of employee satisfaction", Journal of Quality Management, Vol. 6, pp. 29-4	
41	Lee, (1988)	Employee satisfaction is as important as customer satisfaction in influencing organisational performance. Lee (1988) showed that job satisfaction is among the best predictors of turnover. DEV ES
42		Job satisfaction also influences customer perceptions of service quality.
43	Williams (1995); Brown and Mitchell (1993)	Found that employee benefits influence job satisfaction. Indirect costs associated with job dissatisfaction include training recruiting and learning curve inefficiencies, as well as reduction in the client base.
44	Ulmer, D., Syptak, I.M. and Marsland, D.W. (1999),	Employee satisfaction can improve productivity, reduce staff turnover and enhance creativity and commitment. Therefore, employee satisfaction should not be ignored and yet very few businesses seriously consider employee satisfaction.
45	Dubrovski, (2001)	Improving customer satisfaction not only raises company profits, but also facilitates company development.
46	Oshagbemi, (1997b)	Employee satisfaction in the field of higher education
	DEV ES	Oshagbemi (1997b) measured job satisfaction for 566 college teachers, as shown below: ix) teaching;

[		x) research;
		x) administration and management;
A AMA SHELLAR & A SH		xi) present pay;
2		xiii) promotions;
		xiv) supervision/supervisor behaviour;
		xv) behaviour of co-workers; and
		xv) Physical conditions/working facilities.
		Arry Physical conditions working iterates.
47	Comm and Mathaisel, (2000)	Employee satisfaction in the field of private higher education
	DEV ES	Conduct questionnaire surveys on 606 employees of a private higher education organisation to identify the determinants of satisfaction within educational organisations. The findings were as follows:
		v) Workload;
······································		vi) Work atmosphere;
A 19. Sec. 1. 1. A.		vii) Decision-making;
AV THE VALUE		viii) Ethics/fairness;
	UTARI	
48	Fosam, E.B., Grimsley, M.F.J. and Wisher, S.J. (1998)	Employee satisfaction in the field of police organisation
	THE BUDY BASE	Analysed police organisations to find a suitable employee satisfaction model taking the South Yorkshire Police (SYP) as an example. As shown below;
		vii) Customer focus
		viii) Supervision;
		ix) Goals and objectives;
		x) Training and development;
		xi) Pay; and
		xii) Benefits.
49	Kusku, (2001)	Proposed applying employee satisfaction surveys to the
		employees of a Turkish college, and applied the
	DEVEC	following dimensions for measuring their satisfaction:
	DEV ES	
		viii) General satisfaction
		ix) Management satisfaction;
-		x) Colleagues;
		xi) Other working group satisfaction;
		xii) Job satisfaction;
		xiii) Work environment; and
		xiv) Salary satisfaction.

50	Metle, (2003)	Conducted employment satisfaction surveys on female employees in the Kuwaiti public government sector (KGS), and identified the following employment
	DEV ES	<ul> <li>satisfaction factors:</li> <li>vii) Overall job satisfaction</li> <li>viii) Pay and security;</li> <li>ix) Co-workers;</li> <li>x) Supervision;</li> <li>xi) Promotion; and</li> <li>xii) Content of work.</li> </ul>

## 2.8.1.1 Tangibles

This dimension is referring to all equipment and facilities that help to accommodate employees.customers. According to meriam-webster dictionary, tangible means 1) capable of being perceived especially by the sense of touch; palpable; substantially real; material; 2) capable of being precisely identified or realized by the mind; 3) capable of being appraised at an actual or approximate value.

The tangible elements had been divided into physical product which means goods consumed during the service productions process and physical support which explained the framework that enables or facilitates the production of service (Lehtinen and Lehtinen, 1991). In addition, Marshall and Murdoch (2003) described tangible elements as the physical representation or images of service to be provided. Hshiung and Yi (2004), depicted tangible as strongly associating with technology efficiency or reliability and timeless between the tangible elements and the reliability concepts. This is contrast with Parasauraman et.al (1998)

Basically, different industries will use different types of tangibles. According to Panda, T.K., & Das, S (2014), tangibles are defined as aspects of a service that can be 'felt' without actually purchasing the service. It cannot be touched, tasted and be seen. It is the visible aspects of the service that are used by business to improve employee satisfaction.

Many researchers tend to consider tangible element is important. However, this is contrast with Gummesson (1992) which primarily focused on the importance of tangible elements in her work on physical facilities where service is delivered. The surrounding where services were delivered will impact the way people perceived the organisation and their feelings towards the organization.

This statement also was supported by Chowdhary *et. al* (1995) that suggested that Zeithaml *et al.*, (1990) proved that tangible elements of its services and the appearance of employees are important.

# 2.4 Reliability

Reliability refers to the ability to deliver expected standard at all time, how the organization handle employee services problem, perform right services for the first

time, providing services within promised time and maintaining error free record. (Taufik, M., K, N., Adzmir, M., & S, A. 2016)

Yang and Fang (2004), stated that reliability consists of accurate order of fulfilment, accurate record, accurate quote, accurate billing, and accurate calculation of commissions which keep the service promising to the employee.

Singh (2010), found that reliability is the most important aspect. It refers to the three important factors:-

- i) Accessibility. It is the most important aspect which means that the service is available whenever it is required and desired.
- ii) **Continuity**. It means that the service continues to be available and up to the required standard. Maintenance of standard is imperative for adhering to the continuity factor. If the service is not reliable, there is a chance that employee will not trust its reliability and will switch to the other service provider (Zeithaml, 2009).
- iii) **Performance**. It includes high quality service delivery and exceeds the employees' satisfaction thus gaining their loyalty for longer time. Strong

network connection and its availability at all times is significant to maintain high reputation of service (Sutherland, 2007; Sube, 2012)

Marshall and Murdoch (2003) mentioned that reliability is the ability to provide a resolution during the service encounter and as part of delivering the promise from the service provider. As for Gronroos (2001), suggested that reliability is important for employees so that they can rely on the service provider, its employees and the systems created by the service provider.

#### 2.5 Responsiveness

Stands for willingness to help employees and provide prompt service. This dimension emphasizes attentiveness and promptness in dealing with employee's request, questions, complaints and problems. It is also captures the notion of flexibility and ability to customise the service to employee needs. Fitszmmons (2006), suggested that responsiveness is the ability to help employe at the moment of time.

According to Kumar, Mani, Mahalingam, & Vanjikovan (2010), responsiveness consists observing the needs and demands of employees, easy and fast processing time, taking care of employees individually by the staff, solving problems and employees' safety. It is also refers to speed and timeliness of service delivery including willingness and readiness of employees to provide service when a problem occurs and quick problem solving.

Zeithaml *et al.*, (1990) defined responsiveness as the interests shown in providing service to employees when required. It is searched that willingness or readiness of employees to provide the required service without any inconvenience at any time will strongly influence the level of employees' satisfaction (Parasuraman *et al.*, 1988).

2.6 Relationship between Independent Variables and Dependant Variable

2.6.1 Relationships between Tangible and Employees' satisfaction

Quantitative findings by Munisamy, Chelliah and Mun (2010) on service quality delivery and its impact of employee satisfaction in the banking sector in Malaysia revealed that reliability and responsiveness dimension have a relationship but it has no significant effect on employee satisfaction. Only tangible dimension has positive relationship and have significant impact on employee satisfaction. The result analysis is not in congruent to findings by Kitapci, Akdogan and Dortyol (2014) where they found responsiveness dimension have significant relationships with employee satisfaction but not tangibles and reliability. However, result of the study by Shukor, Samiran, Saleh, Hasnan (2015) contradicts with Kitapci *et al.*, (2014) and Parasuraman *et al.*, (1988). Their findings indicated that the only tangible contributed to the employee's satisfaction. Even they found that safety and security is another important dimension which is not highlighted by other researcher.

#### 2.6.2 Relationships between Reliability and Employee Satisfaction

Amin and Issa (2008) found that reliability is the strongest predictor of employees' satisfaction in Malaysian Islamic banks. They indicated that reliability had the highest regression coefficient. This is also in line with Amri (2003) which the result of the study shows that reliability was the most important work factor.

Omar (2015), in his empirical studies found that reliability is the most important dimension of e-commerce quality. In the virtual environment, it is vital to make employees to trust that the organization is going to perform what it is a promise to do. Reliability in his study was representing the ability of the web site to fulfil orders correctly, deliver promptly, and keep personal information secure.

# 2.6.3 Relationships between Responsiveness and Employee Satisfaction

Khfafa and Shafi, (2013) found that responsiveness to be the strongest indicator of employees' satisfaction in Libyan Islamic Banks followed by reliability, empathy and assurance.

Hong & Marimuthu (2014), result of their study shows that responsiveness and employees' satisfaction are positively correlated. The banks in Malaysia prefer a friendly bank where they are willing to assist employees in banking operations. Willingness to assist employees is likely to bring a positive satisfaction of employees in Malaysian banking industry. This is in line with Mengi (2009) which found that responsiveness is related to employees' satisfaction.

However, the above findings are not in line with the study by Othman and Owen, (2001) and Akhtar *et al*, (2012) who stated that compliance is the strongest predictor of employees' satisfaction.

Universiti Utara Malaysia

## 2.6.4 Relationships between Work Factors and Employees' satisfaction

Having consistent work factors that can reduce employees' attraction costs and that can increase sales and market share, is believed to be the best way to maintain employees' loyalty and satisfaction in service industry. Work factors; tangible, reliability and responsiveness are considered as among the most important aspect required for business improvement and to creating competitive advantage (Tsoukatos & Rand, 2007). Previous research has proved that work factor is directly related to employees' satisfaction and loyalty. Therefore, it has been proposed as a key concept for organizations. (Rajaratnam Munikrishnan, Sharif & Nair, 2014; Giovanis, Zondiros & Tomaras, 2013).

Negi (2009), the idea of linking work factors and employees satisfaction has existed for a long time. He is doing a study to investigate the relevance of employeesperceived service quality to measure employees overall satisfaction in the context of mobile services. The findings showed that reliability and network quality are the key factors in evaluating overall work factors. However, tangibles, empathy and assurance should not be ignored when evaluating perceived service quality and employees' satisfaction.

Work factors are the overall investigation of services, whereas employees' satisfaction presents the performance results of service provided to employees (Hossein & Sahel, 2013). Employees' satisfaction is broad concept as compared to work factors. However, these two concepts found to have common points.

Work factors part of employee satisfaction. Work factors focuses on services dimensions. Employee satisfaction more extensive compared to work factors. (Hossein & Sahel, 2013). This is because organisations' survival depends on employees' satisfaction & loyalty. Improving work factors is assumed to be important so as to increase employees' satisfaction.

35

According to Saravanan and Rao (2007), employee satisfaction is based on the level of work factors delivered by the service providers. This is determined by the consumer's cumulative experiences at all of the points of contact with company (Cicerone *et al.*, 2009). This shows that some link between work factors and employees' satisfaction highlights the importance of employees' satisfaction when defining of quality (Wicks & Roethlein, 2009).

Lack of service quality and employee satisfaction by the service providers has been considered the largest determinant in reducing profitability. According to Huang, Wu, Chuang and Han (2014) and Gauri (2013), the trigger of complaint behaviour is due to lack of quality and service failures.

Ali and Raza (2015), findings of the study on employee satisfaction in Islamic Bank in Pakistan confirm that compliance, assurance, responsiveness, tangible, empathy and reliability are distinct constructs. These dimensions have a positive and significant impact on employees' satisfaction However, in general their estimations highlight that they more satisfied with the compliance dimension. They believe that Islamic banks in Pakistan are consistent with sharia principle and the operation is based on sharia compliance.



This chapter encompass the research method used in this study. In this chapter, it will consist of theoretical framework, research design, operational definition, instrumentation, data collection, sampling, data collection procedures and techniques of data analysis.

# 3.2 Research Framcwork

This study is specifically analyzing the relationship between employee satisfaction as a dependant variable and work factors as independent variables which consists of tangible, reliability and responsiveness.





## 3.3 Research Design

A research design is a framework for conducting the research paper which details the procedure and process that necessary to gathered information, data gathering methods, instruments to be used, how instruments will be administered and how information will be organized and analysed (Mondofacto, 2009).

This study is aimed to analyze several variables that influence employee satisfaction among employee of XYZ Berhad. This research specifically analyse connection within the dependant variable which is employees' satisfaction and independent variables which are tangible, reliability and responsiveness. The study comprises of descriptive statistic. Statistical method was used in this study to test and analyze the relationship between both variables. The findings are generated by using statistical analysis and data management system (Mujis & Daniel, 2010).

Printed questionnaire are distributed to all respondents in XYZ Berhad and also staff based in XYZ Headquarters. Sample size for this study is 92 employees from the total population of 120 employees. Sample size is selected from sample table which simplified by Krejcie and Morgan (1970). The data were keyed in into Statistical Package for Social Science (SPSS) 21.0 for analysis purposes.

The main purpose is to identify cause and effect relationship using descriptive, correlation and regression analyses.

# 3.4 **Operational Definition**

There are four variable examined in this study. The four variables are tangible, reliability and responsiveness and employee satisfaction.

#### 3.4.1 Tangible

In this study, tangible referring to physical facilities, equipment, personnel and communication materials such as brochure of promotion product which related to housing loan. While, personnel is refer to a dedicated team namely AP who provide services to employee.

# 3.4.2 Reliability

Reliability in this study referring to ability of the dedicated team who provide services to employees in performing the promised services and how committed they are in doing their job. It is also includes the level of accuracy in delivering information to the employees.

## 3.4.3 Responsiveness

Responsiveness in this study refers to attitude of the dedicated team in performing their task. This includes their willingness to help employee and how prompt they give feedback when the need arises.

# 3.4.4 Employee Satisfaction

Employees' satisfaction in this study refers to the satisfaction of employees towards services provided by XYZ Berhad. Employees of XYZ Berhad are divided into two groups of two different job categories which are executive and non-executive.

# 3.4.4 Eligible Employee

In this study, eligible employee refers to 52 employees based in XYZ headquarters and 40 employees from XYZ Berhad who has applied for housing loan for the year of 2016.

#### 3.5 Measurement of Variables

The measurement is based on the structured questionnaire which was distributed to the respondents. The researcher measures the work factors and employee satisfaction. The questionnaire for this research consists of three sections which are as follows:



In section A of the questionnaire, the nominal scale was used to measure the demographic characteristic of respondents including gender, age, work sector, duration and so on. This section is appropriate to use this scale because the variables measured are discrete and referred to as either categorical or non-metric (Hair *et al*, 2007).

In section B, questions are regarding the work factors; tangible, reliability and responsiveness. A 5-point likert scale was used to measure the respondent's perception regarding the work factors in XYZ Berhad. The respondents indicated

the level of importance of statement with responses that ranged from (1) strongly disagree to (5) strongly agree.

In section C, question regarding overall employee satisfaction on work factors. The questionnaire was designed to examine overall of the employees' satisfaction towards work factors provided by XYZ Berhad. Again, this study used a 5-point likert scale to measure the level of respondents' agreement with statements through their response that ranged from (1) strongly disagree to (5) strongly agree.

According to the Table 3.1, it shows the summary of the sources of variable measurement used for this study. The questionnaires were adapted from the previous study. There was having 18 items which is covered the 4 variables that has been used to achieve the objective of this study. All of the variables were used 5 point likert scale as a variable measurement.

Universiti Utara Malavsia

Summary of Sources Variables	Items	Measurement	Sources
		Scale	
Tangibles	4	Likert (1-5)	Tsang and Qu (2000), Juwaheer and Ross (2003), Zeithmal et. al (1990)
Reliability	5	Likert (1-5)	Sheng-Hsuing and Yi Chun (2003), Nelson and Hailin (2000)

Table 3.1

Responsiveness	4	Likert (1-5)	Juwaheer and Ross (2003), Carev (2008), Zeithmal et. al (1990)
Employee Satisfaction	5	Likert (1-5)	Carev (2008)

# 3.6 Questionnaire Design

There are three sections of questionnaire as shown below:

- i) Section A Demographic Profile of Respondents
- ii) Section B Work Factors; Tangible, Reliability and Responsiveness
- iii) Section C- Employee Satisfaction Measurement

Section A, this questionnaire stated which related to the demographic or respondents personal information. There are some basic of the personal information question were asked about the respondents' demographic profile. It consists of 7 questions

Section B, it is comprises questions on work factors; tangible (4 items) reliability (5 items) and responsiveness (4 items)

Section C, questionnaires consist of 5 items that related to the dependant variables; employees' satisfaction. The following table shows that the summary of completed questionnaire in this research as below:

 Table 3.2

 Summary of the questionnaire

#### Section A

Gender	1	Section A:1
Age	I	Section A:2
Work Sector	]	Section A:3
Job Category	ł	Section A:4
Duration of being XYZ Berhad's employee	1	Section A:5
Frequency of communication	ì	Section A:6
Method of communication	]	Section A:7

## Section **B**

Tangible	4	Section B:1-4
Reliability	5	Section B:1-5
Responsiveness	4	Section B:1-4



# 3.7 Data Collection

The population targeted for this research are the employees from XYZ Berhad and employees based in XYZ Headquarters in year 2016. In this research, the questionnaire applied data analysis method and it were adopted from the previous research which regards related field which have been mentioned above.

Data for this quantitative study was gathered using structured questionnaire which is distributing to the employees of XYZ Berhad and employees based in XYZ headquarters after permission had been gained from the Human Resources Manager. There are two ways that have been applied for this study to collect the data from the respondents. As for employee from XYZ Headquarters, the questionnaires were distributed to the respondents with the assistance of Human Resources Manager of XYZ Headquarters. Meanwhile, for the employee of XYZ Berhad, the questionnaires were distributed by the researcher.

The researcher provided the suitable and sufficient time for the targeted respondents to answer the questionnaire. The researcher has commenced the distribution of questionnaire begun to distribute 92 questionnaires to the respondents on 7 Mac 2017 and respondent's feedback was showed as per table below:-



Ta	ble	3.3	

Statistic Respondents of Employees satisfaction

Category	XYZ HEADQ	UARTER	XYZ BERHAD		
Category	Distributed	Received	Distributed	Received	
Executive	26	26	20	20	
Non-Executive	26	26	20	20	
Total		92			

# 3.7.1 Population and Sampling Design

Group of population chosen in this research were from employees of XYZ Berhad and XYZ Headquarters consists of 92 employees from job category of Executive and Non Executive. Table 3.4 shows the population of the respondents



According Krejcie and Morgan (1970), a minimum sample size of 92 is required to represent a number of 120 populations.

Table 3.5 Table Krejcie and Morgan

<u>M</u>	S	N	Ş	N	Ş	N	\$	N 3	\$
TD ;	10	100	80	280	162	800	260	2600	334
15	14	1 10	86	290	165	850	265	3000	34
20	19	120	92	300	169	900	269	3500	240
25	24	130	97	320	175	950	274	4000	35
30	28	140	103	340	181	1000	278	4500	35
35	32	150	108	360	186	1100	285	5000	35
° <b>40</b> ′′′′ 1		160	113	380	181	1200	291	6000	36
45	4Ô	160	118	400	196	1300	297	7000	36-
50 1	44	190	123	420	201	1400	302	6000	35.
55	48	200	127	440	205	1500	306	9000	364
60	52	210	132	460	210	1600	310	10000	37.
65	56	220	136	480	214	1700	313	15000	374
70	59 59	230	140	500	217	1800	317	20000	
75	63	240	144	550	225	1900	320	30000	374
80	66 56	250	148	600	234	2000	322	40000	ŚŻ
85	70	260	152	650	242	2200	327	50000	38
90 1	73	270	155	700	248	2400	331	75000	382
<u>95</u>	76	270	159	750	256	2600	335	100000	38-

# 3.7.2 Sampling Technique

Sampling design means that the researcher chooses an appropriate number of elements from the whole population of unit analysis. According to Uma Sekaran (2003) the sampling design is important to generalize the total population. As mentioned above, the population size is 120 and sample size was calculated as 92 based on Krejcie & Morgan, 1970.

# 3.7.3 Techniques of Data Analysis

Statistical Package and Social Science (SPSS) 21.0 was used to process raw data. All the completed responses were collected for the analysis. Thereafter, data analysis was carried out including descriptive analysis, reliability analysis, correlation analysis and regression analysis.

## 3.7.3.1 Descriptive Analysis

Descriptive analysis is a discipline that represents the main characteristics and features of a collection data. The objective of the discipline is to summarize a set of collection of data. Aaker, Kumar, and Day (2007) stated that the descriptive analysis is applied to outline and summarizes the key characteristic of data collected from targeted respondents. It includes measurement and identification of central tendency includes the mean test, variability or dispersion which includes standard deviation, range and coefficient of variance.

# 📃 🖉 Universiti Utara Malaysia

Frequency analysis also can be defined as a descriptive statistical method that will present the total number of occurrences of every responded chosen by the targeted respondents. The analysis mostly used to measure and analyse the data of respondents's demographic.

There are seven (7) of demographic information in section A which includes gender, age, work sector, job category, length of service, frequency of communication and method of communication are using frequency analysis. In this study, mean value used to describe the average level of agreement of respondents to items questions.

# 3.7.3.2 Reliability Analysis

The reliability analysis was conducted by computing Cronbach' Alpha for each measure. The reliability of a measure shows that the stability and consistency of the instrument in measuring a concept. It helps to access the goodness of measurement. Cronbach's Alpha is a reliability coefficient that indicates how well the items in a set are positively correlated to one another.

Cronbach's Alpha is computed in terms of the average intercorrelations among the items measuring the concept. The closer the Cronbach's Alpha is to 1, the higher the internal consistency reliability (Sekaran, 2006).

 Table 3.6
 Interpreting the Cronbach's Alpha Value

Degree of Reliability	Cronbach's Alpha		
Excellent	.90 to 1.0		
Good	.80 to .89		
Acceptable	.70 to .79		
Questionable	.60 to .69		
Poor	.50 to .59		
Unacceptable	< .50		

From the Table 3.6, the excellent degree of reliability shows by value of Cronbach's Alpha more than .90, George and Mallery (2003). The range of .80 to .89 is classified as good whereas the range of .70 to .79 is classified as acceptable. Range .60 to .69 is questionable, range .50 to .59 considers as poor and the range .50 is unacceptable.

#### 3.7.3.3 Pearson Correlation Analysis

Zikmund (2003) highlighted that the Pearson Correlation Coefficient is a statistical measure of the strength of a linear relationship two metric variables. So, the relationship between two variables exists and is measured on the same interval or ratio scale.

Thus, Pearson Correlation Coefficient is applied to analyse the relationship between variables and two-tailed significance level is used to null and hypotheses. The coefficient at ranges from +1.0 indicates perfect positive relationship, -1.0 indicates perfect negative relationship. Hair, Bush, & Ortinau (2002), a correlation of 0 means there is no linear relationship between the two variables.

The researcher has shown the measurement scale which was highlighted by the previous research. In this research, the researcher refers to the scale rating that developed by Hair, Money, Samouel and Page (2007) and can be interpret as below:

Table 3.7		
Scale rating developed by	, Hair, Money,	Samouel and Page

The Coefficient Scale	Relationship strength
0.91 - 1.00	Very strong
0.71 - 0.90	Strong
0.41- 0.70	Moderate
0.21 - 0.40	Weak
0.01 - 0.20	Very Weak

## 3.7.3.4 Multiple Regressions

Multiple regression analysis is done to examine the simultaneous effects of several independent variables on a dependant variables that is interval scaled (Cavana *et al., 2001)*. For this study, multiple regression analysis was used to know whether all of the independent variables (tangible, reliability and responsiveness) would significantly add to the variance explained in the dependent variable (employee satisfaction).

According to Pallant (2007), multiple regressions allow prediction of a single dependent continuous variable form a group of independent variables. This analysis leads to answer research questions; Is there any significant relationship between work factors (tangible, reliability, responsiveness) and employee satisfaction. Multiple regressions evaluate which dimension of work factors that influences employee satisfaction.

## CHAPTER 4

## **RESULTS AND DISCUSSION**

#### 4.1 Introduction

This chapter will discuss the result of data analysis that has been acquired from the data collected through questionnaires. All variables used in the study will be analysing by using SPSS software version 21.0. It also aims to answer the research questions and objectives that have been designed in Chapter 1. Besides that, it will include the overview of data collected, reliability analysis and descriptive analysis.

# 4.2 Response Rate

There were 92 sets of questionnaires distributed among respondents which comprise of respondents from XYZ Headquarter and XYZ Berhad itself. All respondents had completed the questionnaires, therefore the percentage of response rate for this research 100%.

## 4.3 **Profile of Respondents**

There were seven demographic factors that have been included in this study; gender, age, job category, work sector, length of service, frequent of communication and also method of communication. There were 52 respondents from XYZ Headquarter and 40 respondents from XYZ Berhad participated in this study.

52

Table 4.1Demographic of respondents

Demographic	Categories	Frequency	Percentage (%)
Gender	Male	48	52.17
Senati	Female	44	47.83
Age	< 20		
5	21 - 30	32	34.8
	31 - 40	35	38.0
	41 - 50	21	22.8
	51 - 60	4	4.3
	> 61	-	-
	XYZ	52	56.52
Work Sector	Headquater		
		40	43.48
	XYZ Berhad		
Job Category	Executive	46	50.0
	Non-Executive	46	50.0
Length of service	ers <mark>zi Utara</mark>	Majays	12.0
	1 - 5	41	44.6
	5 - 10	29	31.5
	> 10	11	12.0
Freenancy of communication	once a week	18	19.6
Freqnency of communication	once a month		
	once a year	15 2	16.3 2.2
	when	57	62.0
	necessary	57	02.0
Method of communication	Telephone	38	41.3
	E-mail	-	-
	Face to face	54	5 <b>8</b> .7
	Others	-	_

#### 4.3.1 Gender



Based on the Figure 4.1, there were 48 male respondents and 44 female respondents from the 92 sets of questionnaires that had distributed. The results showed that among 92 respondents, there were 47.83% were female and 52.17% were male which were majority in the gender category.

## 4.3.2 Age

Figure 4.2 shows that majority of the respondents are at the age 31-40 years old representing 38% of the total number of 92 respondents. This followed by aged 21-30 years old second largest age group with 34.8% which represent 32

respondents. The respondents from age group 41-50 years old contributed 22.8% which were 21 respondents. The respondent from the age group 51-60 years old contributes 4.3% which were 4 respondents.



# 4.3.3 Work Sector



Figure 4.3 Statistic of Work Sector

As shown in Figure 4.3, there are 52 respondents from XYZ Headquarters which represent 56.52% followed by 43.48% respondents from XYZ Berhad which were 40 respondents.



# 4.3.4 Job Category

Figure 4.4 Statistic of Job Category

The above Figure 4.4 shows that 50% of the respondents which representing 52 respondents were from XYZ Headquarters. 26 of respondents are from executive level and the rest of 26 are from non-executive level. As for XYZ Berhad, questionnaire was distributed to the 50% of respondents which represent 40 respondents. 20 respondents are from executive level and the rest of 20 are from non-executive level and the rest of 20 are from non-executive level.

# 4.3.5 Length of Scrvice



Figure 4.5 Statistic of Length of Service

Figure 4.5 shows that majority of the respondents having been an employee of XYZ Berhad for about 1-5 years representing 44.6% of the total number of 92 respondents. This is followed by 5-10 years second largest duration of being XYZ Berhad's employee group with 31.5% which represent 29 respondents. 11 respondents with 12% being a XYZ Berhad's employees for more than 10 years and the rest of 11 respondents which represents 12% used to be XYZ Berhad's employees for one year and less.

# 4.3.6 Frequency of Communication

Figure 4.6 shows that majority of respondents communicate with XYZ Berhad's staff when necessary only which represent 57 respondents with 62%. This is followed by 18 respondents with 19.6%. Only 15 respondents prefer to communicate with the staff of XYZ Berhad once in a month which represent 16.3% and the rest of 2 respondents which represent 2.2% only communicate once a year.



Figure 4.6 Frequency of Communication

## 4.3.7 Method of Communication



Figure 4.7 above shows that 54 of respondents which represent 58.7% prefer to deal face to face with the staff of XYZ Berhad when the need arises and the rest of 38 respondents which represents 41.3% prefer communicate using telephone.

# 4.4 Descriptive Analysis

The analysis of descriptive statistic is used to determining the level of service quality in XYZ Berhad. It will presents in term of mean and standard deviation. Total average mean will be divided into 3 categories; high, moderate and low according to the mean score as stated in the Table 4.2 below.

# Table 4.2Category of Mean Score

Number	Range	Category	
1	1-2.33	Low	
2	2.34 - 3.67	Average	
3	3.68 - 5.00	High	

Table 4.3Descriptive Statistic of Tangibles

Tangible	N	Mean	Std.	
			Deviation	
Q1 : XYZ Berhad uses modern looking equipment	92	3.92	.650	
Q2 : XYZ Berhad has physical facilities	which are	Malaysi	a	
visually appealing	92	3.89	.582	
Q3 :XYZ Berhad employees' appearance looking	are neat 92	3.83	.745	
Q4 : XYZ Berhad's material (brochure) associated are visually appealing	92	3.76	.617	
To	tal Average Mea	n 3. <b>8</b> 5		

Table 4.3 displays mean and standard deviation for tangibles. From the study, it shows that the highest mean of tangibles which are 3.92 and standard deviation of .650 that refers to question 1. It shows that most of the respondents agree that XYZ
Berhad uses modern looking equipment. The lowest mean is 3.76 and standard deviations of .617 refer to question 4. Overall, score mean for tangible of service quality is high with 3.85.

	NI	Maara	Std.	
Reliability	N	Mean	Deviation	
1 : XYZ Berhad issues error-free bills, statement's and other documents	92	3.96	.717	
Q2 : XYZ Berhad fulfils when it promises to do something by a certain time	92	3.73	.626	
Q3 : XYZ Berhad's employees tell you exactly when services will be performed	92	3.86	.698	
Q4 : XYZ Berhad provides on-time service	92	3.94	.652	
Q5 : XYZ Berhad performs the service right the first time	92	3.90	.663	
Total Aver	age Mean	3.87		

Table 4.4Descriptive Statistic of Reliability

Table 4.4 displays mean and standard deviation for reliability. From the study, it shows that the highest mean of reliability which are 3.96 and standard deviation of .717 that refers to question 1. It shows that most of the respondents agree that XYZ Berhad issues error-free bills, statements and other documents. The lowest

mean is 3.73 with standard deviation .626 that refer to question 2. Overall, score mean for reliability of service quality is 3.87 which fall under category high.

			Std.
Responsiveness	Ν	Mean	
			Deviation
Q1 : XYZ Berhad's employees give you prompt service	92	3.97	.628
Q2 : XYZ Berhad's employees are always willing to help you	92	4.08	.622
	92	3.84	.627
Q3 : XYZ Berhad's employees are never too			
busy to respond to requests			
	92	3.92	.578
Q4 : XYZ Berhad's employees understand			
your specific needs			
Total Average Mean		3.95	

Table 4.5Descriptive Statistic of Responsivenes.

Universiti Utara Malaysia

Table 4.5 displays mean and standard deviation for responsiveness. From the study, it shows that the highest mean of responsiveness which are 4.08 and standard deviation of .622 that refers to question 2. It shows that most of the respondents agree that service provider willing to help their employees. The lowest mean is 3.84 and standard deviation of .627 that refer to question 3. Overall, score mean for responsiveness of service quality is high with 3.95 and it is proved that majority of staff's who provide services are helpful, committed and always ready to co-operate.

	'NT	Mean	Std.	
Employee Satisfaction	N	мени	Deviation	
Q1:Overall, XYZ Berhad has an updated technology	92	3.88	.676	
Q2 : I'm satisfied with the performance of XYZ Berhad	92	4.03	.686	
Q3: I will use facilities provided by XYZ Berhad in future	92	4.08	.567	
Q4: Overall, i do believe that XYZ Berhad has a minimal error level	92	3.89	.636	
Q5 : Physical conditions of tangible features	92	4.03	.582	
Total Average Mo	ean	3.98	.,	

Table 4.6 above indicates the descriptive analysis of employee satisfaction towards services provided by services provider. According to the above table, there are 5 items was used to evaluate employee satisfaction of XYZ Berhad. The highest mean is 4.08 that refer to question 3, whereby the lowest mean is 3.88 that refer to question 1. Overall, the answer range of respondents is agreed that they will use facilities provided by XYZ Berhad in the future.

# 4.5 Reliability Analysis

In this study, several items were measured for each variable. The Cronbach's Alpha statistics were used to test internal consistency or reliability of group items.

Construct	Items	Cronbach's Alpha
Tangibles	4	0.799
Reliability	5	0.902
Responsiveness	4	0.768
Employee satisfaction	4	0.802

Table 4.7 *Reliability for Variables* 

Universiti Utara Malaysia

According to George and Mallery (2003), the range of Cronbach's Alpha value are divided into 6 categories; range of .80 to .89 is classified as good, range of .70 to .79 is classified as acceptable,.60 to .69 is questionable, range .50 to .59 considers as poor and the range .50 is unacceptable. Table 4.9 below shows degree reliability of service quality based on the value of cronbach's alpha.

Table 4.7, shows the value of Cronbach's Alpha for every variable was ranging from 0.768 to 0.902 where the value is closer to 1. It shows that the results that all construct of main variable are reliable and exceeded 0.7. Based on the reliability

analysis, reliability indicates the highest alpha coefficient 0.902 which achieved excellent level.

This is followed by employees' satisfaction which produced alpha coefficient 0.802 and achieved good results. Meanwhile, both tangible (0.799) and responsiveness (0.768) achieved acceptable results.

Table 4.8Interpreting the Cronbach's Alpha Value

Degree of Reliability	Cronbach's Alpha
Excellent	.90 to 1.0
Good	.80 to .89
Acceptable	.70 to .79
Questionable	.60 to .69
Poor	.50 to .59
Unacceptable	<.50

# 4.6 Pearson Correlation Testing

Table 4.9 Correlation Coefficient				
<u><u>v</u>v</u>	1	2	3	4
1. Tangibles	1			
2. Reliability	.761**	1		
3. Responsiveness	.580**	.709**	1	
4. Employee Satisfaction	.644**	.753**	.705**	1

\*\*Correlation is significant at the 0.01 level (2-tailed)

In this research, the researcher applied Pearson Correlation Coefficient to analyse the relationship between two variables with two-tailed significance level. Based on the Table 4.9, it shows that all of the independent variables are positively correlated with the dependant variable at significant level 0.01.

The researcher used scale rating developed by Hair, Money, Samuel and Page (2007) and the results shows that the highest and strongest correlation is between employee satisfaction and reliability of service quality (r= 0.753, p < 0.01), followed by responsiveness of service quality (r=0.705) with moderate relationship. The lowest correlation is tangibles of service quality (r = 0.644, p < 0.01) which shows moderate relationship with employee satisfaction.

In summary, the result found that all variables have positive correlation between each of the variables.

eveloped by Hair, Money, Samo	
The Coefficient Scale	Charles Relationship strength
 0.91 - 1.00	Very strong
0.71 - 0.90	Strong
0.41 - 0.70	Moderate
0.21 - 0.40	Weak
0.01 - 0.20	Very Weak

# 4.7 Multiple Regression Coefficient

A multiple regression was conducted to test the hypotheses. The regression analysis was used to examine the dimensions of relationship between the independent and independent variables on this research. There are three independent variables, namely tangibles, reliability and responsiveness. Those variables are predicted to influence the dependant variable, which are employees' satisfaction. The result of multiple regressions is presented in the Table 4.12.

Table 4.12 *Multiple Regression* 

Model	Standardized Coefficients	Т	Sig.
	Beta		
(constant)		3.098	.003
Tangibles	.138	1.378	.172
Reliability	.412	3.572	.001
Responsiveness	.334	3.635	.000

# a. Dependent Variable: Employee satisfaction

		Model Su	ummary <sup>b</sup>	
Model	R	R Square	Adjusted R	Std. Error of the Estimate
1	.796 <sup>a</sup>	.634	.621	1.45191

a. Predictors: (Constant), Responsiveness, Tangibles, Reliability

b. Dependent Variable: Employees' satisfaction

The result of the analysis explained that the value of the R square is 0.634 or 63.4%. The variance of 63.4% explains the contribution of the independent variables on the dependent variable. Table 4.12 also indicates that two independent variables, namely reliability ( $\beta$ =0.412, p<.01) and responsiveness ( $\beta$ =0.334, p<.01) were significantly related with the dependent variable while, reliability ( $\beta$ =0.412) is the most influential independent variables. However, dimension of tangibles

( $\beta$ =0.138) shows non-significant result. The results reveal that employees satisfied with the level of reliability and responsiveness provided by XYZ Berhad.

As per Santhiyavalli (2011), reliability and responsiveness are the most important dimensions considered by employee while evaluating the overall work factors. This is also in line with the research conducted by Arsali and Smadi (2005) who suggests that reliability factor has the highest impact on employees' satisfaction (Shala & Pira, 2017)

#### 4.8 SUMMARY

This chapter explain details about the results and findings of this study. It is also discuss on response rate, descriptive analysis, reliability analysis, correlation analysis and regression analysis. All the information results were analysed using SPSS 21.0 version. The findings reveal that reliability and responsiveness positively related and has a significant relationship whereas tangible is non-significant.

# **CHAPTER 5**

# DISCUSSION, CONCLUSION AND RECOMMENDATION

## 5.1 Introduction

This final chapter of this thesis provide summary of this study. Its discuss implications of this thesis and summarizing the results. It is also deliberated and summarized significance of the result in study, discussions about the outcomes of the findings obtained in chapter four, suggested future research direction and implication of the study.

# 5.2 Summary of Study

This section presented based on analysis of findings in order to answer research objective one, research objective two and research objective three. This study will contribute to the growing literature by examining the relationships between work factors namely tangible, reliability and responsiveness and employees' satisfaction. The correlation and multiple regressions were conducted in determining level of employees' satisfaction towards work factors provided by XYZ Berhad.

# 5.3 Discussion

5.3.1 Research Objective 1: Relationship between tangible and employee satisfaction

Based on the correlation analysis, it showed that tangible of service quality has a moderate relationship to the employee satisfaction. It has been identified as the lowest important indicator as compared to other two (2) independent variables; responsiveness and reliability. Meanwhile, regression analysis reveals that tangible is not significantly correlated to employees' satisfaction.

This is in line with the result analysis of study by Kitapci, Akdogan and Dortyol (2014) where they found that responsiveness dimension have significant relationships with employees' satisfaction. While, the rest of the other two work factors namely; tangibles and reliability has not significantly related to the dependent variable. Khamis (2014), the result analysis in his study also reveal that tangibles are not significantly predictor to employees' satisfaction.

However, result study by Shukor, Samiran, Saleh, Hasnan (2015) contradicts with Kitapci *et al* (2014) and Parasuraman *et al* (1988) where their findings indicated that employees' satisfaction is contributed by tangible. Even they found that safety and security is another important dimension not highlighted by other researcher.

As for this study, we can summarized that tangible is not a main factor that contributed to the dependent variable.

**5.3.2 Research Objective 2:** Relationship between reliability and employee satisfaction

Based on the correlation analysis, it showed that reliability has a strong relationship to the employees' satisfaction and its has been identified as the most important indicator as compared to other two (2) independent variables; responsiveness and tangible. While, regression analysis showed that reliability was significantly correlated to employees' satisfaction. In addition, the reliability was found as most dominant independent variable as compared to other independent variable.

This study is congruent with result of the study conducted by Amin and Issa (2008) which found that reliability is the strongest predictor of employees' satisfaction in Malaysian Islamic banks. This finding was supported by Al-Tamimi and Al-Amri (2003), which showed that reliability, is the most important in work factors.

**5.3.3 Research Question 3:** Relationship between responsiveness and employee satisfaction

Based on the correlation analysis, it showed that responsiveness has a strong relationship to the employees' satisfaction. Meanwhile, regression analysis found that responsiveness was significantly correlated to employees' satisfaction.

This result of the study is congruent with Khafafa and Shafi (2013) which showed that responsiveness is the strongest indicator of employees' satisfaction. In addition, study from Hong & Marimuthu (2014) and Mengi (2009), also reveal that responsiveness and employees satisfaction are positively correlated.

# 5.4 Implication of the study

Most of the literatures related to employees' satisfaction involving the independents variables selected supports that the theory that this variables will significantly have relationship to employees' satisfaction. However, after research being done, it reveals that one the selected independent variable does not have significant relationship with the employees' satisfaction that is tangible. Meanwhile, the other two variables that are reliability and responsiveness has a significant relationship and regards as the most important factors that contribute to the level of employees' satisfaction in XYZ Berhad.

The findings of this study also have significant implication for policy makers. The management of XYZ Berhad should re-examine their policies which related to the housing loan. The existing polices and procedure should be re-designed and improvised in line with the current situation.

#### 5.5 Recommendation

There are many ways to improve employees' satisfaction. According to Wright, Gardner, Moynihan & Allen, 2005, in order to improve employees' satisfaction, it is important to measure and establish the existing levels first.

However, due to its multi-faceted nature, the measurement of employee satisfaction varies from one organisation to the other. Some organisations use anonymous employees' satisfaction surveys which are administered periodically to measure the levels of employee satisfaction (Deshpande, Arekar, Sharma & Somaiya, 2012).

In other organisations, meetings are held between management and small groups of employees where the latter are asked questions pertaining to their satisfaction (Ybema, Smulders & Bongers, 2010).

As quoted by Schulz, 2001, exit interviews are the primary employees' satisfaction measurement tools. The importance of these methods lies in that they elicit satisfaction sentiments from employees themselves (Schneider, Hanges, Smith & Salvaggio, 2003). Employees' satisfaction has thus been widely recognised as a predictor of productivity and performance in organisations (Dawal, Taha & Ismail, 2009; Silvestro, 2002).

#### 5.5.1 Maintaining the level of reliability

XYZ Berhad should maintain or control their level or reliability. Reliability means the trust that gain from employees or other people. Level of attentiveness and promptness in dealing with employee's request, questions, complaints and problems should be improved. Thus, when organization gain trusty from employees, organization must take care about their product and services in order to keep employee always trust organization.

#### 5.5.2 Technology

As for the operation side, the management of XYZ Berhad should review modus operandi of their business. It is a must for them to make sure their operations are supported with technology. Based on the data gathered for this research, the researcher realised that the most preferable method of communication in XYZ are using phone and dealing face to face. Thus, to increase level efficiency of service; management should initiate new method of communications, i.e., e-mail and other social media such as Facebook. Mariapan (2006), stated that revolution of information technology has brought astonishing changes in business environment which no other sector has been influenced by advancement in technology as much as banking and financial institution

#### 5.5.3 Upgrade Website

Management of XYZ Berhad need to upgrade and fixing up website accordingly. The contents of the website must be up-to-date and the websites must be link to social media. Forbes in their survey found that 78% of consumers said that post that made by the organisation on social media influence their purchases. The website also must be mobile friendly or using friendly search engine to avoid website traffic. Meanwhile, as one of the strategy to compete with competitors, management of XYZ must ensure their website as good as their other competitors, at least at par or better than the other banks outside. In term of brochure, besides maintain the printed brochure, upgraded website also can be regard as an online brochure. It is undeniable that upgraded website are more encouraging as well as facilitate interaction between management and employee. Even, complaints/request by employees can be done through online and can be entertain immediately. It is automatically can improve the effectiveness level of communication in XYZ Berhad.

Universiti Utara Malaysia

# 5.5.4 Educate the employee

Educate dedicated team (AP) is an ongoing process. In order to gain employee trust and confident, XYZ Berhad must provide an educated of AP team. Educated team is believed more committed in carry out their duty and will treat employees nicely. They must be given on job training. This is to make sure that they have good product knowledge, understand their benefits accurately and persuasively. The fact is, employees more respond to enthusiastic dedicated team that are passionate about their products and services and eager to share the benefits with them.

# 5.5.5 Handling Employee Complaint

All the complaints must be properly recorded and entertain within seven working days. There must be a one box of employee complaint especially for walk-in employee based in headquarters. So, they can easily voice out their dissatisfaction as well as giving suggestion to the company. An authorised dedicated team should be appointed to entertain the complaints which responsible to read all the complaints and take the necessary actions. They should understand on the basic procedure in complaints handling procedure as follows:-



Creating task force to settle the crucial cases is a compulsory when the said cases are not settled at the level of lower level staff. In this case, upper level manager especially under operation side should be made known about the current issues. Their involvement will help faster and tightened the process. The management should re-examine the existing complaints handling policy. It should be improvised in line with the current need.

# 5.5.6 Improving Service quality

Service quality is important in organization. In order to increase profit, organization must give high level of service to employee. XYZ Berhad must continuously improvise their service quality from time to time. By doing that, XYZ Berhad was able to maintain the level of service quality. Improving service quality can be improved by focusing on employee problems and to evaluate the effect of each on employee satisfaction.

# 5.6 Direction for Future Research

This study was directed to examine the relationship between work factors and employee satisfaction in XYZ Berhad. The result from this study was drawn from a small sample size of 92 respondents. Therefore, a larger sample size is desirable for the future research. The current study was a cross sectional study but to determine the causal path of the study variables a longitudinal study would have been more appropriate (Poo, 2004). Therefore, other variables like corporate image which is can influence employee satisfaction will also be tested on the future study. Then including some of the moderator factors and look forward to direct or indirect relationship towards employee satisfaction also can be made in the future research.

# 5.7 Conclusion

In summary, from this study it shows that the applications of work factors; tangible, reliability and responsiveness are useful method in examining level of employee satisfaction towards service quality of XYZ Berhad. With this finding, the manager was able to figure out the important element of service quality that effecting their employee satisfaction.

Universiti Utara Malaysia

#### REFERENCES

- A Qualitative Case Study on Customer Satisfaction with Low-Cost Airlines: Perceptions of Airline Customers Dissertation Manuscript Submitted to Northcentral University Graduate Faculty of the School of Business Technology Management In Partial Fulfillment of the Requirements for the Degree of DOCTOR OF PHILOSOPHY by BOSTON AGOWEI EDOGI Prescott Valley, Arizona February, 2017. (2017).
- Abdullah, Musa, Zahari, Rahman, & Khalid. (2011). The Study of Employee Satisfaction and its Effects Towards Loyalty in Hotel Indutsry in Klang Valley, Malaysia. International Journal of Business and Social Science, 2(3) Special Issue (January).
- Aaker, D. A., Kumar, V., & Day, G. S. (2007). *Marketing research* (9 ed.). Massachusetts: Wiley.
- Adil, M., Falah, O., Al, M., & Albkour, A. M. (2013). SERVQUAL and SERVPERF: A Review of Measures in ServicesMarketing Research, 13(6).
- Aff, J. B. F., & Felix, R. (2017). Business & Financial Affairs Service Quality and Customer Satisfaction in Selected Banks in Rwanda, 6(1), 1–11. http://doi.org/10.4172/2167-0234.1000246
- Ahmad Fiaza Abdul Shukor, Maslina Samiran, Hasan Saleh & Norlena Hasnan (2015), The Relationship between Service Quality, Customer Satisfaction and Customer External Complaints Intentions in Commercial Parking Facilities in Klang Valley, Malaysia, International Journal of Science Environment, 4 (3), 595-615.
- Albinsson, M. (2006). Customer satisfaction and retention: the experiences of individual employees. http://doi.org/10.1108/09604520410513668
- Ali, M., & Raza, S. A. (2017). Service quality perception and customer satisfaction in Islamic banks of Pakistan: the modified SERVQUAL model. *Total Quality Management & Business Excellence*, 28(5-6), 559-577. http://doi.org/10.1080/14783363.2015.1100517
- Allan, M. M. (2016). The Relationship Between Service Quality and Customer Satisfaction and Retention in Ghana's Luxury Hotels, XV(4), 60-84.
- Al-Tamimi, H.A.H. and Al-Amiri, A. (2003), "Analysing service quality in the UAE Islamic banks", Journal of Financial Services Marketing, Vol. 8 No. 2, pp. 119-32

- Amin, M., Aldakhil, A. M., Wu, C., Rezaei, S., & Cobanoglu, C. (2017). The structural relationship between TQM, Employee Satisfaction and Hotel Performance. *International Journal of Contemporary Hospitality Management*, 1256 – 1270.
- Amin, M., & Isa, Z. (2008). An examination of the relationship between service quality perception and customer satisfaction A SEM approach towards Malaysian Islamic banking, 1(3), 191-209. http://doi.org/10.1108/17538390810901131
- Arslan, M., Iftikhar, M., & Zaman, R. (2014). Effect of Service Quality Dimensions on Customer Satisfaction: A Comparative Analysis of Pakistan Telecom Sector, 15(4).
- Artar, O. Ki., Okumuş, Ş., & Genç, E. G. (2016). ASSESSING CUSTOMER AWARENESS AND SELECTION CRITERIA OF ISLAMIC AND CONVENTIONAL BANKS IN 2. The Turkish Participation Banking Industry, 17(2), 255–271.
- Azeem Khattak, N. (2010). Customer satisfaction and awareness of Islamic banking system in Pakistan. African Journal of Business Management, 4(5), 662–671. Retrieved from http://www.academicjournals.org/AJBM
- Azman, I., Ilyani Ranlan, R., & Nur Afiqah, F. (2016). Service quality as an antecedent in enhancing customers ' behavioural intentions : A case study of Malaysian army medical centers. GEOGRAFIA OnlineTM Malaysian Journal of Society and Space, 12(2), 179–190.
- Burke, R.J., Graham, J. & Smith, F. (2005). Effects of reengineering on the employee satisfaction-customer satisfaction relationship. *The TQM Magazine*, 17: 358-363.
- Campus, c., & Lumpur, k. (2015). the relationship between service quality, customer satisfaction and customer external complaints intentions in commercial parking facilities in klang, 4(3), 595-615.
- Cavana, R.Y., Delahaye, B.L., & Sekaran, U., (2001). Applied Research; Qualitative and quantitative method, Sydney; John Wiley & Sons Inc.
- Chi, C.G. & Gursoy, D. (2009). Employee satisfaction, customer satisfaction, and financial performance: an empirical examination. *International Journal of Hospitality Management*, 28: 245-253.
- Chowdhory, N & Monika P. (1995), "Prioritizing Service Quality Dimensions", Journal of Service Marketing .Vol.2, No. 1, pp.124-I30

- Cronin, J.J. and Taylor, S.A. (1992), "Measuring service quality: a reexamination and extension", Journal of Marketing, Vol. 6, July, pp.55-68
- D, E. C. P. (2010). MEASURING CUSTOMER SATISFACTION: MUST OR NOT?, 6(2), 76–88.
- Dabholkar, P. A., Shepherd, C. D., & Thorpe, D. I. (2000). A Comprehensive Framework Service Quality: An Investigation of Critical Conceptual and Measurement Issues Through a Longitudinal Study. *Journal of Retailing*, 76(2), 139–173.
- Dharmalingam, S., Ramesh, R., & Kannan, K. V. (2011). Investigating the Service Quality Dimensions to Customer Satisfaction and Loyalty of New Private Sector Banks in Tamilnadu –An Empirical Study. *Research journal of Social Science and Management*, 37-46.
- Donnelly, M. (2009), Building customer sense of fulfillment: A customer experience based approach in a tourism context. Bachelor of Business Studies in Marketing, (June), 205-212.
- De Matos, C. A., Henrique, J. L., & de Rosa, F. (2009). The different roles of switching costs on the satisfaction-loyalty relationship. *International Journal of Bank Marketing*, 27, 506-523. doi:10.1108/02652320911002331
- Demirbag, M.,Koh, S.C.L., Tatoglu, E. & Zaim, S. (2006). TQM and market orientation's impact on SMEs' performance. *Industrial Management & Data Systems*, 106(8): 1206-28.
- Elizabeth, A. (2013). 2013 service quality factors on customers satisfaction in syarikat bekalan air selangor (syabas) the infulence of servce quality factors on customer satisfaction and drinking water quality in syarikat bekalan air selangor (syabas) master of science management universiti utara malaysia (uum) july 2013, (july).
- Ellickson, M.C., & Logsdon, K., (2001). Determinants of job satisfaction of municipal government employees. State & Local Government Review, 33(3), 173-184.
- Emrah Cengiz Ph. (2010). Measuring Customer Satisfaction : Must or Not ? Journal of Naval Science and Engineering, 6(2), 76–88.
- Eshghi, A., & Roy, S. K. (2008). SERVICE QUALITY AND CUSTOMER SATISFACTION : AN EMPIRICAL INVESTIGATION IN INDIAN MOBILE

TELECOMMUNICATIONS SERVICES, (1), 119–144.

- Fitszmmon, J.A.,&Fitszmmon, M.J. (2006), Service Management: Operations, Strategy, Information Technology. United States of America: McGraw-Hill Higher Education. Graetz,
- Ganguly, R. (2010). Quality of work life and job satisfaction of a group of university employees. *Asian Journal of Management Research*, 1(1), 209–216.
- Gauri, V. (2013). Redressing Grievances and Complaints Regarding Basic Service Delivery. World Development, 41, 109–119. [22]
- George, D., & Mallery, P. (2003). SPSS for windows step by step: A simple guide and reference (4th ed.). Boston: Allyn & Bacon.
- Giese, J., & Cote, J. (2000). Defining consumer satisfaction. Academy of Marketing Science Review, 1(1), 1–22.
- Giannikis, S.K., & Mihail, D.M. (2011). Modeling job satisfaction in low-level jobs: Differences between full-time and part-time employees in the Greek retail sector. *European Management Journal*, 29(2), 129–143.
- Goldman, N. (2011). drivers of member advocacy and detraction in credit unions : key contributors to net.
- Gregory, K. (2011). *The Importance of Employee Satisfaction*. Retrieved from <u>http://www.neumann.edu/about/publications/NeumannBusinessReview/journ</u> <u>al/Review2011/Gregory.pdf</u>
- Grönroos, C. (2001). The Perceived Service Quality Concept A Mistake? *Managing Service Quality*, 11(3), 150–152. http://doi.org/10.1108/09604520110393386
- Hair, J., Bush, R., & Ortinau, D. (2002). *Marketing research: Within a changing information environment* (2nd ed.). McGraw-Hill Education.
- Hair, Jr., J. F., Money, A. H., Samouel, P. & Page, M. (2007). Research methods for business. Chichester: John Willey & Sons Ltd.

Hansemark, O. C. & Albinson, M. (2004) "Customer Satisfaction and Retention:

*The Experiences of Individual with Employees*", Managing Service Quality, Vol. 14 (1)

- Hernon, P., & Nitecki, D. a. (2001). Service Quality: A Concept Not Fully Explored. Library Trends, 49(4), 687–708. http://doi.org/10.5539/ass.v8n13p201
- Hong, T. L., Marimuthu, M., Tunku, U., & Rahman, A. (2014). relationship between service quality and customer satisfaction: a study of malaysian banking industry, 2(2), 53-72.
- Hsu, S.H. & Wang, Y.C. (2008). The development and empirical validation of the employee satisfaction index model. *Total Quality Management & Business Excellence*, 19: 353-366.
- Hung, Y. H., Huang, M. L., & Chen, K. S. (2003). Service quality evaluation by service quality performance matrix. *Journal of Total Quality Management*, 14(1), 79–89. http://doi.org/10.1080/1478336032000044843
- Homburg, C., Giering, A. and Hentschel, F., 2013. Der Zusammenhang zwischen Kundenzu-friedenheit und Kundenbindung. In: Bruhn, M.; Homburg, C., Handbuch Kundenbindungsmanagement: Grundlagen, Konzepte, Erfahrungen. 8th ed. Wiesbaden: Springer Gabler.
- Hossein, V., & Sahel, F. (2013). Factors influencing customer satisfaction with the success factors identified in the insurance industry. *African Journal of Business Management, Vol.* 7(21), 2026-2032.
- Ihtiyar, A. (2015). The Role of Intercultural Communication Competence on Service Reliability and Customer Satisfaction, 5(1).
- Isberg, S. (2010). Using the SERVQUAL Model to assess Service Quality and Customer.

Ilies, R., Wilson, K.S., & Wagner, D.T. (2009). The spillover of daily job satisfaction

onto employees' family lives: The facilitating role of work-family integration. Academy of Management Journal, 52(1), 87–102.

Irving, P.G., & Montes. S.D. (2009). Met expectations: The effects of expected and delivered inducements on employee satisfaction. *Journal of Occupational* and Organizational Psychology, 82(2), 431–451.

- Islam, M.Z., & Siengthai, S. (2009). Quality of work life and organisational performance: empirical evidence from Dhaka Export Processing Zone. Paper presented at the ILO Conference on 'Regulating for Decent Work', held at the International Labour Office, Geneva, July 8–10.
- Ismail, A., Sufardi, Y., & Yunan, M. (2016). LogForum SERVICE QUALITY AS A PREDICTOR, *12*(4), 269–283. http://doi.org/10.17270/J.LOG.2016.4.7
- Jayaraman Munusamy, Shankar Chelliah and Hor Wai Mun, " Service Quality Delivery and Its Impact on Customer Satisfaction in the Banking Sector in Malaysia," *International Journal of Innovation, Management and Technology* vol. 1, no. 4, pp. 398-404, 2010.
- Julie Pallant. SPSS Survival Manual, A step by step guide to data analysis using SPSS for windows (Version 12). Allen & Unwin.
- Jung, H.S. & Yoon, H.H. (2015). The impact of employees' positive psychological capital on job satisfaction and organizational citizenship behaviors in the hotel. *International Journal of Contemporary Hospitality Management*, 27: 1135-1156.
- Karatepe, O.M., Uludag, O., Menevis, I., Hadzimehmedagic, L. & Baddar, L. (2006) The effects of selected individual characteristics on frontline employee performance and job satisfaction. *Tourism Management*, 27: 547-560.
- Kenya Projects Organisation Building Capacities. (2012, August 25). Retrieved from http://www.kenpro.org/sample-size-determination-using-krejcie-and-morgan table
- Kitapci, O., Akdogan, C., & Dortyol, İ.T. (2014). The Impact of Service Quality Dimensions on Patient Satisfaction, Repurchase Intentions and Word-of-Mouth Communication in the Public Healthcare Industry. *Procedia - Social and Behavioral Sciences*, 148, 161-169.
- Khafafa, A. J., & Shafii, Z. (2013). Measuring the Perceived Service Quality and Customer Satisfaction in Islamic Bank Windows in Libya Based on Structural Equation Modelling (SEM), 2, 56–71.
- Koonmee, K., Singhapakdi, A., Virakul, B., & Lec, D. (2010). Ethics institutionalization, quality of work life, and employee job-related outcomes: A survey of human resource managers in Thailand. *Journal of Business Research*, 63(1), 20-26.
- Kotler, P., & Gertner, D. (2002). Country as brand, product, and beyond: A place marketing and brand management perspective. *Journal of Brand Management*,

9(4), 249-261. http://doi.org/10.1057/palgrave.bm.2540076

- Kumar, S. A., Mani, B. T., Mahalingam, S., & Vanjikovan, M. (2010). Influence of Service Quality on Attitudinal Loyalty in Private Retail Banking: An Empirical Study. *IUP Journl of Management Research*, 9(4), 21-38.
- Kumra, R., 2008, "Service Quality in Rural Tourism: A Perspective Approach". Conference on Tourism in India-Challenges Ahead, Idia, P. 424-431
- Krajnakova, E., Navikaite, A., Navickas, V. (2015), Paradigm Shift of Small and Medium-Sized Enterprises Competitive Advantage to Management of Customer Satisfaction, *Inzinerine Ekonomika-Engineering Economics*, 2015, 26(3), pp. 327-332.
- Lambert, L.S., Edwards, J.R., & Cabic, D.M. (2003). Breach and fulfillment of the psychological contract: A comparison of traditional and expanded views. *Personnel Psychology*, 56(4), 895–934.
- Lee, J.S., Back, K.J. & Chan, E.S. (2015). Quality of work life and job satisfaction among frontline hotel employees: a self-determination and need satisfaction theory approach. International Journal of Contemporary Hospitality Management, 27: 768-789.
- Lee, M. C. (2003). An Examination of the Quality and Context-Specific. *Journal of* Service Research, 5(4), 345–355. http://doi.org/10.1177/1094670503251135
- Lehtinen, U. & Lehtinen, J.R. (1991), "Two Approaches to Service Quality Dimensions", *The Service Industries Journal*, Vol.11, No.3, pp.287-303.
- Liang, D. (2008). The Determining Factors of Customer Loyalty for Luxury Hotels in US. *The Journal of International Management Studies*, 3(2), 167–175.
- Liniere, M. De, Committee, R., Chairperson, C., Member, C., Officer, C. A., & Riedel, E. (2013). Walden University.
- Lu, C., Berchoux, C., Marek, M. W., Chen, B., Lu, C., Berchoux, C., ... Christian, C. Y. (2015). Service quality and customer satisfaction : qualitative research implications for luxury hotels. http://doi.org/10.1108/IJCTHR-10-2014-0087

- Lu, H., While, A.E., & Barriball, K.L. (2005). Job Satisfaction among Nurses: A Literature Review. *International Journal of Nursing Studies*, 42(2), 211–227.
- Marshall, G. & Murdoch, I. (2003), "Service Quality in the Marketing of Consulting," *Journal of Marketing*, Vol.52.337-352
- Mahamad, O., & Ramayah, T. (2010). Service quality, customer satisfaction and loyalty: a test of mediation. *International Business Research*, 3(4), 72.
- Martin, M. J., & Hays, F. (2016). CUSTOMERS 'DETERMINATION OF SERVICE QUALITY AND SATISFACTION IN A RETURN / REPAIR PROCESS: A QUANTITATIVE STUDY, 20(1), 36–53.
- Mediating role of customer satisfaction on service quality and customer loyalty in grocery store in shopping malls in saudi arabia doctor of philosophy universiti utara malaysia september 2014. (2014), (september).
- Mohsan, F., & Nawaz, M. M. (2011). Impact of Customer Satisfaction on Customer Loyalty and Intentions to Switch: Evidence from Banking Sector of Pakistan, 2(16), 263-270.
- Mosavi, S. A., & Ghaedi, M. (2012). A survey on the relationships between customer satisfaction, image, trust and customer advocacy behavior. African Journal of Business Management, 6(8), 2897–2910. http://doi.org/10.5897/AJBM11.1465
- Moyes, G. D., & Shao, L. P., Newsome, M. (2008). Comparative analysis of employee job satisfaction in the accounting profession. *Journal of Business* & Economics Research, 6(2), 65-81.
- Mutlaq, N., & Otaibi, A. (2014). An Overview of Customer Loyalty, Perceived Service Quality and Customer Satisfaction: Brief on Saudi Grocery Stores, I(1), 79–122.
- Mikulic, J. and Prebežac, D. (2011), "A critical review of techniques for classifying quality attributes in the Kano model", *Managing Service Quality*, Vol. 21 No. 1, pp. 46-66.

Muijs. D. (2004). Doing Quantitative Research in Education with SPSS. SAGE

Publications Ltd.

- Namukasa, J. (2013). The influence of airline service quality on passenger satisfaction and loyalty: The case of Uganda airline industry. *The TQM Journal*, 25(5), 520–532. http://doi.org/10.1108/TQM-11-2012-0092
- National Business Research Institute. (n.d.). Retrieved from https://www.nbrii.com/customer-survey-white-papers/10-factors-that-affectcustomer-satisfaction/.Education, I. N. (n.d.). *quantitative research*.
- Negi ,R. (2009). Determining customer satisfaction through perceived service quality: A study of Ethiopian mobile users. International Journal of Mobile Markeiting, 4(1), 31-38.
- Nippatlapalli, A. R. (Nov Dec 2013). A Study On Customer Satisfaction of Commercial Banks: Case Study on State Bank of India. *IOSR Journal of Business and Management (IOSR-JBM)*, 15(1): 60 - 86.
- Omar, H. F. H. (2015). Determining the Influence of the Reliability of Service Quality on Customer Satisfaction: The Case of Libyan E-Commerce Customers, 5(1), 86-89. http://doi.org/10.5296/ijld.v5i1.6649
- On, S., Development, T. H. E., Quality, O. F., Models, M., Steering, F. O. R., Services, B., Customer, T. O. (2017). Study on the development of quality measurements models for steering business services, 19(44), 95–110.
- Ooi, K.B., Lee, V.H., Chong, A.Y.L. & Lin, B. (2013). Does TQM improve employees' quality of worklife? Empirical evidence from Malaysia's manufacturing firms. *Production Planning & Control*, 24: 72-89.
- Okumus, H. S., & Genc, E. G. (2013). Interest free Banking in Turkey : A study of Customer Satisfaction and Bank selection . *European Scientific Journal edition* vol.9, No.16, 144-166.
- Oliver, R. L. (2010). Satisfaction: A behavioral perspective on consumer (2ndEd.). Armonk, NY: Sharpe.Othman, A. and Owen, L. (2001), "Adopting and measuring customer service quality (SQ) in Islamic banks: a case study in Kuwait Finance House", International Journal of Islamic Financial Services, Vol. 3 No. 1, pp. 1-26.

Panda, T. K., & Das, S. (2014). The Role of Tangibility in Service Quality and Its

Impact on External Customer Satisfaction : A Comparative Study of Hospital and Hospitality Sectors.

- Parasuraman, A., Valarie, A., & Leonard, L. (1988). Servqual: A Multiple-Item Scale For Measuring Consumer Perceptions of Service Quality. *Journal of Retailing*, 64(1), 12–40. http://doi.org/10.1016/S0148-2963(99)00084-3
- Parasuraman, A., Zeithaml, V. A., & Berry, L. L. (1985). A conceptual model of service quality and its implications for future research. *Journal of Marketing*, 49(4), 41–50. http://doi.org/10.2307/125143
- Parasuraman, A., Zeithaml, V. A., & Berry, L. L. (1988). Servqual: A Multiple-Item Scale For Measuring Consumer Perc. *Journal of Retailing*, 64(1), 12.
- Peiro, J. M., Ramos, J., & Sa, R. M. (2009). Testing a hierarchical and integrated model of quality in the service sector: functional, relational, and tangible dimensions, 20(11), 1173–1188. http://doi.org/10.1080/14783360903247577
- Prajogo, D.I. and Sohal, A.S. (2004), The multidimensionality of TQM practices in determining quality and innovation performance: an empirical examination, *Technovation*, Vol. 24, pp. 443-453.
- Price, J.I. (2001). Reflections on the determinants of voluntary turnover. International Journal of Manpower, 22(7), 660-624.
- Raab, G., Werner, N., 2009. Customer Relationship Management: Aufbau dauerhafter und profitabler Kundenbeziehungen.3rd ed. Frankfurt am Main: Verlag Recht und Wirtschaft GmbH
- Rahman, M. S., Khan, A. H., & Haque, M. (2017). A Conceptual Study on the Relationship between Service Quality towards Customer Satisfaction: Servqual and Gronroos 's Service Quality Model Perspective, 8(13), 201–210. http://doi.org/10.5539/ass.v8n13p201
- Rogelberg, S.G., Allen, J.A., Shanock, L., Scott, C. and Shuffler, M. (2010), Employee satisfaction with meetings: a contemporary facet of job satisfaction, *Human Resource Management*, Vol. 49, pp. 149-172.
- Saravanan, R., & Rao, K.S.P. (2007) .The impact of total quality service age on quality and operational performance: an empirical study. *The TQM Magazine*,

19 (3), 197 - 205

- Sempane, M.E. Rieger, H.S., & Roodt, G. (2002). Job satisfaction in relation to organisational culture. *Journal of Industrial Psychology*, 28(2), 23–30.
- Seth, N., & Deshmukh, S. G. (2005). Service quality models : a review (Vol. 22). http://doi.org/10.1108/02656710510625211
- Shala, E., & Pira, B. (2017). The Evaluation of Service Quality in the Growing Banking Sector in Kosovo, 7(1), 19–24.
- Shaw, J.D., Delery, J.E., Jenkins, G.D. Jr and Gupta, N. (1998), "An organizationallevel analysis of voluntary and involuntary turnover", Academy of Management Journal, Vol. 41 No. 5, pp. 511-25.
- Singh, H. (2006). The Importance of Customer Satisfaction in Relation to Customer Loyalty and Retention by Harkiranpal Singh May 2006 UCTI Working Paper. UCTI Working Paper, (May), 6.
- Sousa-Poza, A. and Sousa-Poza, A.A. (2000), "Well-being at work: a cross-national analysis of the levels and determinants of job satisfaction", *The Journal of Socio Economics*, Vol. 29, pp. 517-538
- Studies, S. (2014). Evolution and Conceptual Development of Service Quality in Service Marketing and Customer Satisfaction, (2008), 1189–1198.
- Suhartanto, D. (2011). An Examination of Brand Loyalty in the Indonesian Hotel Industry. *Brand*, 168.
- Susan M Heathfield, Employee Satisfaction, retrieved from About.com on 23 Nov. 2011<u>http://humanresources.about.com/od/employeesurvey1/g/employee\_satisfy.htm</u>
- Tabaku, E., & Cerri, S. (2016). an assessment of service quality and customer satisfaction in the hotel sector elvira tabaku, 480–489.
- Taufik, M., K. N., Adzmir, M., & S. A. (2016). Customer Satisfaction on Reliability and Responsiveness of Self Service Technology for Retail Banking Services. *Procedia Economics and Finance*, 37(16), 13-20.

http://doi.org/10.1016/82212-5671(16)30086-7

- Tsoukatos, E., & Rand, K. (2007). Cultural influences on service quality and customer satisfaction: Evidence from Greek insurance. Manag. Serv. Qual. 17(4), 467-485.
- Tsui, A.S., Pearce, J.L., Porter, L.W. and Tripoli, A.M. (1997), Alternative approaches to the employee-organization relationship: does investment in employees pay off?, Academy of Management Journal, Vol. 40 No. 5, pp. 1089-121.
- Uma Sekaran. Research Methos for Business A Skill Building Approach, Fourth Edition. John Wiley & Sons (ASIA) Pte Ltd.
- UNTARU, E. N., ISPAS, A., & DAN, I. (2015). Assessing the Quality of Banking Services using the SERVQUAL Model. *Romanian Journal of Marketing*, *2*, 84–92.
- Vijayanand, S., & Selvaraj, M. (2015). Impact of Service Quality on Customer Perception : An Empirical Study in India 's Post Office Savings Banking Sector Using SERVQUAL, 5(3), 189–206.

Universiti Utara Malavsia

- Yang,Z.,& Fang, X. 2004. Online Service Quality Dimensions and Their Relationships with Satisfaction: A Content Analysis of Customer Reviews of Securities Brokerage Services. *International Journal of Service Industry Management*, 15(3), 302-326.
- Wagner, R.; Herter, J. K. 2006. 12: The elements of great managing. New York: Gallup Press.
- Wicks, A. M., & Roethlein, C.J. (2009). A satisfaction Based Definition of Quality Journal of Business & Economic Studies, Vol. 15, No. 1, Spring 2009, 82-97
- Woods, C. (n.d.). *Study.Com What is Employee Satisfaction*. Retrieved from <u>http://study.com/academy/lesson/what-is-employee-satisfaction-examples-factors-quiz.html</u>.

Yuksel, Atila, Yuksel, Fisun, Bilim, yasin, "Destination attachment: Effects on

customer satisfaction and cognitive, affective and cognitive loyalty", *journal of tourism management*, 2010, 31, pp:274-284

- Zhou, L. (2004). A dimension-specific analysis of performance-only measurement of service quality and satisfaction in China's retail banking. *Journal of Services Marketing*, 18(7), 534–546. Retrieved from http://10.0.4.84/08876040410561866
- Zikmund, W. G., Babin, B. J., Carr, J. C., & Griffin, M. (2010). Business research methods (8th ed.). South-Western, Ohio: Cengage Learning.

